# Why Do SSI and SNAP Enrollments Rise in Good Economic Times and in Bad?

Matthew S. Rutledge and April Yanyuan Wu

Center for Retirement Research at Boston College Hovey House 140 Commonwealth Ave Chestnut Hill, MA 02467 Tel: 617-552-1762 Fax: 617-552-0191 http://crr.bc.edu

The research reported herein was pursuant to a grant from the U.S. Social Security Administration (SSA), funded as part of the Retirement Research Consortium (RRC). The findings and conclusions expressed are solely those of the authors and do not represent the views of SSA, any agency of the federal government, the RRC, or Boston College. The authors would like to thank Rebecca Cannon, Natasha Orlova, Patrick McBee, Matthew Lempitsky and Kendrew Wong for research assistance. All errors are their own.

© 2013, Matthew S. Rutledge and April Yanyuan Wu. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

#### Why Do SSI and SNAP Enrollments Rise in Good Economic Times and in Bad?

#### Abstract

The number of participants in the Supplemental Security Income Program (SSI) and the Supplemental Nutrition Assistance Program (SNAP) skyrocketed during the Great Recession. But more surprising is that caseloads for both programs increased during the preceding expansion and during the nascent recovery period after the Great Recession. Using both administrative program data and the Survey of Income and Program Participation (SIPP), this project investigates the persistent growth in SSI and SNAP since 2000. Whereas the existing literature on program caseloads in the post-welfare reform era generally excludes the elderly from the analysis, this project is the first to investigate differences in elderly and non-elderly caseloads, allowing for differential responsiveness over time. Preliminary estimates suggest that the correlation between SSI and SNAP caseloads and economic well-being, and separately caseloads and health, grew stronger over this time. Coupled with a poverty rate that did not fall along with the unemployment rate, and an increase in the share of the population reporting poor or fair health, these correlations helped lead to caseloads that remained roughly constant (SSI) or even increased (SNAP) during the most recent expansion, rather than falling as expected. The increases in caseloads stem from both increases in the entry rates among the newly eligible particularly those in poor health – and decreases in exit rates among low-income beneficiaries.

#### **1. Introduction**

During the Great Recession, participation in SSI and SNAP rose to record levels. But rising caseloads have also occurred when the economy was not in recession. Between fiscal years 2003 and 2007, SNAP caseloads increased by 24 percent, from 21 million to 26 million, the first time in the program's history that caseloads increased during a period of economic growth. During the same period, SSI applications increased, and about 500,000 people were added to the rolls (Figure 1). Although SSI eligibility depends on a less cyclical factor – disability status – the pattern in 2003-2007 runs counter to the late 1990s expansion when SSI applications and awards fell. Further, SNAP and SSI caseloads continued to increase after June 2009, the official end point of the Great Recession and the beginning of modest economic growth.

While a rich literature has documented the relationship between macroeconomic conditions and welfare caseloads, few studies have focused on the unexpected increases in SSI and SNAP caseloads in 2003-2007. The most common method of analysis is a dynamic model of state caseload levels in which current levels are predicted by a program's caseload level in the previous year, the unemployment rate, and employment per capita (Grogger 2003; Klerman and Haider 2004; Schmidt 2012). Studies also include policy parameters as explanatory variables to disentangle the effects of the economy from the effects of policy changes on welfare caseloads. The majority of these studies find that the unemployment rate and other macroeconomic conditions account for a large share of the change in caseload levels. For example, Stapleton, Coleman, and Dietrich (1995) report that a 1-percentage-point increase in the unemployment rate leads to a 2-percent increase in SSI caseloads one year later. Ziliak, Gunderson, and Figlio (2003) estimate that the same change in the unemployment rate leads to a 2.3 percent increase in SSI caseloads one year later. Ziliak, Gunderson, and Figlio (SNAP) caseloads. If these historical relationships had held during the recent expansions, the SSI (SNAP) caseloads should have fallen by about 2.8 (3.2) percent during the mid-2000s economic expansion, instead of increasing by more than 5 (24) percent.

One plausible explanation for the unexpected growth of SSI and SNAP caseloads is the 1996 welfare reform. Given some substitutability between SSI/SNAP and Temporary Assistance for Needy Families (TANF), potential TANF recipients may switch to alternative safety-net programs because of lifetime limits imposed by welfare reform and stringent work

2

requirements.<sup>1</sup> Strict time limits may also increase reliance on SNAP. However, increases in SSI applications and SNAP caseloads during the mid-2000s expansion are observed not only among adults with children, but among the elderly, a group for which TANF should have the least impact (Figure 2). The magnitude of the increase among the elderly is comparable to other age groups. Further, the persistent increase in overall SSI caseloads has masked a more complex relationship: rather than rising, first-time awards have been flat or even decreased for some age groups in the most recent decade. This suggests that factors other than welfare reform played a role.

Using both administrative program data and the *Survey of Income and Program Participation* (SIPP), this paper investigates the continuing growth in SSI and SNAP since 2000 at both the state- and individual- levels. At the state-level, the project investigates which factors impact SNAP caseloads and SSI caseloads, applications, and first-time awards using state panel data, exploiting both across- and within-state variations over time. While much of the previous work focuses on the SSI rolls - that is, the stock of participants – not as much attention has been paid to application and first-award levels – that is, the flows into and out of the program. The distinction between stocks and flows is critically important here, as caseloads will be slow to respond to the business cycle, but applications and awards can respond more rapidly (Grogger 2003; Klerman and Haider 2004, Wu 2009, Schmidt 2012). We examine how each factor contributes to the changing program rolls, including economic conditions, demographic and health variables, and policy variables. To understand the counter-cyclical growth of SSI and SNAP, we examine how the effect of each factor has changed over time.

At the individual-level, the study focuses on the dynamics of public program participation and decomposes the caseload into specific mechanical components for each program, including changes in the number of people who are eligible, and take-up and exit rates among the eligible, to determine which components are responsible for the increased caseloads. The project then examines the direction and magnitude of the effect on each of these components of demographic and economic conditions, program policies, and other policy variables.

<sup>&</sup>lt;sup>1</sup> The populations served by the SSI, SNAP, and TANF programs have historically been similar in terms of observable characteristics, such as sporadic employment history and low educational attainment. In addition, TANF recipients have high rates of physical and mental disabilities (Danziger et al. 2000; Nadel, Wamhoff, and Wiseman 2003/2004).

While the existing literature on program caseloads in the post-welfare reform era generally excludes the elderly from the analysis, this project specifically investigate differences between the elderly and the non-elderly, allowing for differential responsiveness over time. The SSI caseload of elderly individuals has increased over the past two decades, but first-time awards have been flat or even declined, emphasizing the importance of decomposing the changing caseload by age.

The state-level results find several reasons that SNAP caseloads increased during the mid-2000s expansion. First, the positive correlation between a state's poverty rate and its nonelderly SNAP enrollment has grown stronger over time. Since the poverty rate did not fall with the unemployment rate during the economic recovery, the increased responsiveness to the poverty rate contributes to the rising caseload. Second, as the share of the state's non-elderly population reporting fair or poor health has trended upward, the positive correlation between the proportion in fair or poor health and the non-elderly caseload has also grown stronger. Further, the correlation between elderly SNAP enrollment and this health measure has become weaker over time, while the proportion of the elderly reporting fair or poor health, when coupled with changes in the mean values of these variables, offer an explanation for the continuing growth of SNAP since 2000, including expansionary periods.

For the SSI program, the rich data allow us to explore both stocks (caseloads) and flows (applications and first-time awards) at the state level. SSI enrollment is actually negatively correlated with the unemployment rate over the full period, but this relationship has grown less negative over time and even turned positive during the Great Recession, just as unemployment rates spiked upward. As with SNAP, the positive correlation between non-elderly SSI caseloads and poor health grew stronger, and the correlation between elderly SSI caseloads and poor health became weaker and even negative. For SSI applications, we find that increasing responsiveness to poor health and a weakening correlation with the unemployment rate help explain the unexpected growth in SSI applications during the expansion and continuing upward trend since 2000.

Our individual-level analysis indicates there are two primary drivers of the continuing growth in SNAP and SSI since 2000: a fall in the rate at which low income participants who remain eligible leave the programs and a rise in the rate at which newly eligible individuals

4

reporting poor health enter the programs. These factors were also likely to be behind the unexpected increase in SSI and SNAP caseloads specifically during the 2003-2007 economic expansion.

The onset of a new economic expansion should allow for public program budgets to recover from their increased outlays during the preceding recession. But recent expansions suggest that the budgetary burden may not ease, as SSI and SNAP caseloads continued to grow even as the unemployment rate fell. Our study suggests that further reductions in SNAP and SSI caseloads likely will require not only a declining unemployment rate, but commensurate improvements in poverty rates and health for a low-income population that is only tenuously attached to the labor force.

The paper proceeds as follows. Section 2 briefly outlines the SNAP and SSI programs and reviews the existing literature. Section 3 describes the data and sample construction. Section 4 discusses empirical methods and Section 5 summarizes the results, followed by concluding remarks in Section 6.

## 2. Background

*The Supplementary Nutrition Assistance Program (SNAP)*. SNAP is the largest nutrition program for low-income Americans and a mainstay of the federal safety net. In fiscal year 2011, the program served an average of 44.7 million people per month and paid out over \$71.8 billion in benefits (USDA 2012). To receive SNAP, households must meet three financial criteria: a gross-income test, a net-income test, and an asset test.<sup>2</sup> *Gross income* is defined as the total income for all household members, including earnings, investment, and transfers, but excludes most noncash income and in-kind benefits. The gross income limit is set at 130 percent of the poverty line (\$1,640 per month for fiscal year 2012 for a two-person household). *Net income* is then computed by allowing for various deductions from the household's gross income, with the net income limit set at 100 percent of the poverty line (\$1,261). The asset limit in 2012 was \$2,000. A household is automatically or "categorically" eligible for SNAP through the receipt of SSI, the Temporary Assistance for Needy Families (TANF), or General Assistance programs.

<sup>&</sup>lt;sup>2</sup> Under SNAP rules, a household is defined as individuals who share a residential unit and purchase and prepare food together.

Eligibility rules for households with an elderly (age 60 and over) or a disabled member are more liberal than for the rest of the population. First, these households are exempt from the gross income test, and the net income test is more generous by removing the shelter deduction cap and by allowing out-of-pocket medical expenses in excess of \$35 per month per household to be deducted. Second, the asset limit increases from \$2,000 to \$3,250.

The amount of SNAP benefit that a household receives is equal to the maximum benefit level less 30 percent of the household's net income (reflecting that an average household will spend approximately 30 percent of its net income on food). In 2012, an eligible two-person household could receive SNAP benefits between \$16 and \$367 each month.<sup>3</sup>

*The Supplementary Security Income Program (SSI).* Designed to provide financial support to low-income blind, disabled, and elderly individuals, the Supplemental Security Income (SSI) program is currently the largest federal means-tested cash assistance program in the U.S.<sup>4</sup> Enacted in 1972, the SSI program has seen tremendous growth over time, with the number of recipients growing from 4 million in 1974 to over 8 million in 2012. Combined with the SNAP, SSI is intended to raise an elderly household's resources to approximately the poverty line.

The federal SSI program provides a guaranteed income to all eligible individuals. In 2012 the income guarantees were 674 (1,011) per month for a single individual (couple) living in his own home. These amounts are reduced by one-third if the recipient lives in someone else's home and are adjusted yearly for inflation. The SSI benefit is the difference between the income guarantee and their countable income. Countable income disregards the first 20 of unearned income (e.g., Social Security benefits among the elderly), the first 65 of earned income, and one-half of additional earned income per month.<sup>5</sup>

An asset test is also required for participation in SSI. To be eligible for SSI benefits, individuals must have countable assets of less than \$2,000, and couples must have less than \$3,000. With respect to the determination of countable assets, the disregards are substantial.

<sup>&</sup>lt;sup>3</sup> For more details about SNAP and SSI eligibility, refer to Coe and Wu (2013).

<sup>&</sup>lt;sup>4</sup> In 20128, federal payments under SSI totaled \$52.0 billion, compared to just \$6.4 billion in federal assistance payments made under TANF.

<sup>&</sup>lt;sup>3</sup> If there is less than \$20 unearned income, additional earned income can be disregarded. Other disregards are irregularly or infrequently received income of less than \$20 per month, home energy assistance payments, tuition benefits, disaster relief, and the value of SNAP benefits.

Most importantly, owner-occupied housing regardless of value and the value of one car up to \$4,500 are excluded.

Individuals younger than 65 must meet the income and asset tests, and be determined to be unable to work for at least 1 year due to a medical impairment.<sup>6</sup> For the aged (65 and over), individuals are eligible if they meet the income and asset tests, without any health requirement. In addition to the federal program, states have the option of offering supplemental SSI benefits. In 2012, 30 states offered supplements to disabled individuals (or couples) living independently, and a total of 45 states offered at least some form of supplemental benefits, which can be substantial. For example, the income guarantee for a couple living in California in 2012 is \$1,407 (\$396 above the federal level), while in New York the income guarantee is \$1,115. If a state is willing to administer its own program, it is free to alter the eligibility requirements as it wishes, including imposing more (or less) stringent income and asset tests. While federal benefits are indexed for inflation, state benefits are not.

*Literature Review.* The fact that participation in SNAP and SSI behaved contrary to expectations during recent economic expansions has prompted studies to seek alternative explanations for the surprising increase. Mabli et al. (2009) find that the unemployment rate remained a strong predictor of SNAP caseload changes from 2000 to 2008, but its ability to explain the percentage change in SNAP caseloads was small relative to prior studies. The paper also finds that an increase in the participation rate among those eligible explains the increase in caseload of the SNAP during the early 2000s recovery period. They attribute the increase in the participation rate to changes in the unemployment rate and changes in SNAP policies. Further, Mabli and Ferrerosa (2010) report that economic factors, including the unemployment rate, labor force participation rate, minimum wage, and characteristics of the low-wage labor market, explain 55 percent of the increase in SNAP caseloads from 2000 to 2008, and changes in policy factors, including offering broad-based categorical eligibility, program outreach expenditures, and the length of recertification periods, explain 20 percent of the changes over this period. Johnson (2011) explores the cause of SNAP caseload changes in 2000 and finds that a fall in exit rate is likely to be the primary cause of the increase SNAP rolls. But Johnson's work does not

<sup>&</sup>lt;sup>6</sup> The disability definition and determination process is identical to that of the Social Security Disability Insurance (SSDI) program.

provide conclusive evidence on what is responsible for the decline in the exit rate. Bitler and Hoynes (2010) find that the cyclicality of SNAP has increased since welfare reform, as low income families aim to offset reduced TANF benefits.

While the literature has found that SSDI caseloads increase with the unemployment rate (Autor and Duggan 2003, Black et al. 2002), these same studies find that SSI is comparatively less cyclical. Schmidt (2012) points out that SSI caseloads should be expected to have a weaker correlation with macroeconomic conditions; SSDI has a requirement related to both total and recent work experience, so SSDI applicants are much more likely to have been recently employed than applicants to SSI, which has no work requirements. As a result, the correlation between unemployment rates and SSI activity are subject to ongoing debate, with some studies finding that applications and awards increase with the unemployment rate (Rupp and Stapleton 1995, Stapleton et al. 1998, Stapleton et al. 1999, Coe et al. 2011), while others find a negative correlation with labor market conditions (Beatty and Fothergill 1996 and 2002, Garrett and Glied 2000, Schmidt and Sevak 2004).

Most relevant to our SSI state-level analysis is Schmidt (2012), which examines the determinants of growth in SSI caseloads across states and over time. The study finds that economic conditions and welfare reform significantly affect SSI participation, and since welfare reform the responsiveness of the SSI program to business cycles has grown stronger. But Schmidt (2012) focuses on the impact of welfare reform on the SSI caseload, rather than the unexpected caseload growth in the mid-2000s, and does not distinguish stocks from flows.

## 3. Data and Sample

We use both aggregate and individual level data from 1996 through 2011, with special focus on the period between the peak unemployment rate in June 2003 and the beginning of the recession in December 2007.

The state-level analysis exploits the panel-data structure of state data, to determine what factors affect SSI and SNAP caseloads. We use the official monthly estimates of state SSI caseloads, applications, and first-time award levels reported by the Social Security Administration (SSA) and SNAP caseloads reported by the U.S. Department of Agriculture. Given that a number of studies have documented significant underreporting of program receipts in large national surveys (Marquis and Moore 1990, Bollinger and David 1997, Bilter et al. 2003,

8

Meyer and Sullivan 2008, Meyer et al. 2009), making use of administrative data improves our estimation.

There are a number of reasons that we expect to see state-level variation in SSI and SNAP participation. First, while economic conditions and the underlying health of the population vary dramatically by state, the eligible populations for SSI and SNAP also vary geographically. Second, states differ in program generosity and the stringency of disability determinations (Maestas et al. 2010 for an example).

One potential concern is that SSI caseloads may not differ between states, or across regions. Schmidt (2012) shows sufficient variation across a small number of states, even within the same region. Figure 3 shows that this variation extends to all states. Similar patterns are observed for SSI applications (Figure 4), first-time awards (Figure 5), and the SNAP program as well (Figure 6).

The individual-level analysis makes use of the 1996, 2001, 2004, and 2008 Survey of Income and Program Participation (SIPP) panels, excluding children under 18. The SIPP is a nationally-representative longitudinal survey of households conducted by the U.S. Census Bureau. The main objective of the SIPP is to provide comprehensive information about income and program participation of individuals and households in the United States. Every four months over a two- to four-year period, respondents are asked a battery of questions on their labor market participation, sources of income, demographics and family structure, wealth, and public program participation during each month between interviews.

#### 4. Empirical Strategy

*State-level Analysis.* Adapting the models estimated by Blank (2001) for AFDC and Schmidt (2012) for SSI, we estimate the correlation between state-level SNAP and SSI activity and key variables:

$$C_{st} = \beta E_{st} + \alpha P_{st} + \gamma D_{st} + S_s + \delta_t + \varepsilon_{st} \quad (1)$$

where *C* is the outcome of interest: the proportion of adults in the relevant age group that are receiving benefits from (separately) SNAP or SSI (i.e., the caseload) in state *s* in year *t*, or the number of SSI applications or first-time awards divided by the age-appropriate population in state *s* in year *t*. *E* is a vector of economic variables; *P* is a set of policy variables, and *D* is a

vector of demographic and health characteristics. S is a set of state controls and  $\delta$  is a set of indicator variables for years 1996-2011 to control for nationwide economic changes in any given year. We estimate separate regressions for the elderly and non-elderly.

The set of economic variables includes the three-year moving average of the year-to-year change in hourly wages at the 10<sup>th</sup> percentile of the state's wage distribution; this variable attempts to capture the influence of slow earnings growth, or even decline, for those near the lower end of the income distribution (Leonesio and Del Bene 2011). This set also includes the concentration of manufacturing and the share of the service industry, to capture the nature of employment. To determine the correlation with the poverty rate, we include the proportion of the entire population under the federal poverty line. These economic variables are calculated from the Annual Social and Economic Supplement of the *Current Population Survey* (the March CPS). We also include the average out-of- pocket medical expenditure by state, from the Centers for Medicare and Medicaid Services; medical expenditures have grown considerably faster than inflation (Smith, Newhouse, and Freeland 2009) and at differential rates across states (Collins 2011), increasing the incentive to apply for SSI for its associated Medicaid coverage.

To measure the cyclicality of each public program, we include the annual unemployment rates by state from the Bureau of Labor Statistics' *Local Area Unemployment Statistics*. The degree to which the unemployment rate sufficiently characterizes the labor market environment of individuals deciding whether to participate in the program has been largely ignored in the literature. Given the fact that over 79 percent of SNAP participants were out of the labor force in 2008 (Wolkwitz and Trippe 2009), in sensitivity tests, we use non-employment rate to pick up any discouraged worker-effect.

Demographic variables analyzed include the proportion of a state's population age 60 or older, the fraction male, the fraction black, the proportion with less than a high school degree, the share of newly-arrived immigrants, and the share of households headed by a single mother. These demographic variables derive from the March CPS.

The primary data for state-level health characteristics are the Center for Disease Control's *Behavioral Risk Surveillance Survey* (BRFSS).<sup>7</sup> Two health variables from the BRFSS are included in our analysis: the proportion of the state judging themselves to be in fair or poor

<sup>&</sup>lt;sup>7</sup> The BRFSS has been administered since 1984 and is the largest ongoing telephone survey in the United States, interviewing 350,000 adults per year about health and health-related behaviors. These series are updates of the variables used by Coe et al. (2011), and we thank the authors for sharing the data.

health, and the share with a self-reported body mass index (BMI) of at least 25, which previous literature has shown to be associated with a higher disability rate (Coe et al. 2011).

Program policies that changed since the early 2000s may impact the caseload. The 2002 Farm Bill gave states much more flexibility over the eligibility requirements for their SNAP program. Increasing the certification period length is one of such change. States have the option of determining how long a household is certified to receive SNAP. A household must recertify their eligibility to keep receiving benefits at the end of its certified period. Certification periods became longer starting in 2003, and by the beginning of the Great Recession, most states were assigning certification periods of 12 months or longer. Increasing the certification period length may increase caseloads because it lowers the participation cost as well as allowing no-longereligible households to keep receiving their benefits longer. We use SNAP Quality Control Data (SNAP QC) to produce a variable measuring the share of SNAP recipients having a certification period of less than 3 months by state-year.<sup>8</sup> Further, we also control for each state's payment error rate, which proxies for the program administration and may affect program participation; this data also comes from SNAP QC database.

Another SNAP policy variable that may impact caseloads is the expanded categorical eligibility. States can choose to offer optional expanded categorical eligibility, which makes eligible any household that receives benefits or service through programs that are at least 50% funded by TANF or maintenance of effort sources. For many of these services the only requirement for eligibility is to have income less than 200% of the poverty line, which is higher than 130% requirement for SNAP gross income eligibility. Expanded categorical eligibility may affect SNAP caseloads by increasing the share that is eligible.<sup>9</sup> Finally, the SSI analysis includes the maximum SSI state supplement for a disabled individual.

Other policy variables analyzed include variables that approximate the relative benefit generosity of other programs. Since previous research suggests that there is some degree of substitutability between SSI/SNAP and TANF (Schmidt and Sevak 2004, Pavetti and Kauff 2006), we control for the maximum TANF benefit for a family of three. We also include the

<sup>&</sup>lt;sup>8</sup> The SNAP QC database is an edited version of the raw data file generated by the SNAP Quality Control System and contains demographic, economic, and SNAP eligibility information for a nationally representative sample of approximately 50,000 SNAP households. The main purpose of the QC review is to assess the accuracy of eligibility determinations and benefit calculations and to determine each state's payment error rate. These data also serve as an important source of detailed demographic and financial information on a large sample of active SNAP participants. <sup>9</sup> We use Fitzpatrick's measure of expanded categorical eligibility, which comes from a database constructed by Mathematica for Food and Nutrition Service.

maximum state Unemployment Insurance monthly benefit to capture potential substitution between UI and SSI/SNAP (Linder, 2011; Coe et al. 2013). As Coe et al. (2011) suggest that state policies regarding access to, and the price of, alternative sources of health insurance impact the decision to apply to SSDI and SSI for the Medicare and Medicaid benefits, respectively, we also control for health insurance regulations, defining a state as strictly regulated if it had both guaranteed issue and some form of community rating. As states with a Medicaid buy-in program provide less strict earnings qualifications for Medicaid eligibility to disabled individuals who work, we also include an indicator variable for a state having a Medicaid buy-in program. We also include an indicator for whether a state has a Republican governor; Coe et al. (2011) finds that SSDI application rates (in particular, concurrent SSDI and SSI applications) are significantly lower with a Republican governor, and Schmidt (2013) shows that SSI caseloads are more cyclical in states with a Democratic governor.

Finally, we interact dummies for time period (2000-2003, 2004-2007, and 2008-2011, with 1996-1999 as the omitted condition) with the main variables of interest to examine how the associations between these variables and SSI/SNAP caseloads have changed over time.

Table 1 presents the descriptive statistics of the state-level data. We have 816 state-level observations, which represent data from 1996-2011 for 50 states plus Washington D.C. On average, 10 percent of the population receives SNAP per year, and 3 percent receives SSI. About 0.8 percent applies for SSI each year, among which 47 percent are awarded benefits. These rates vary widely among states: as many as 28 percent of a state's residents (Oregon in 2011), and as few as 4 percent (Wisconsin in 1999), receive SNAP benefits in a given year.

Measures of demographics, health, and economic conditions also vary considerably. For example, unemployment is 5.7 percent on average, but varies between 1.4 percent and 15.0 percent. The poverty rate varies between 4.5 percent to 25.5 percent, with a mean of 12.4 percent. Between 1.2 and 27.8 percent of the state is employed in manufacturing job. Self-reported poor or fair health ranges from 8.2 percent to 25.4 percent.

State policies potentially influencing public program caseloads also vary by state. About 12 percent of state-years were under a strict health insurance regulating regime, and states moved both into and out of this category during 1996 to 2011. About half of states have the Medicaid buy-in policy. Slightly more than half of the governors were Republican during this period.

12

Maximum UI benefits vary from \$151 to \$629, and maximum TANF benefits for a household of three range from \$120 to \$923 per month.

In terms of public program policies, about 22 percent of state-years have expanded categorical eligibility for the SNAP program, and 8 percent of the population has a certification period of less than three months. The average payment error rate is 6.9 percent. Further, there is a wide variation in SSI maximum state supplement, ranging from \$0 to \$520.

*Individual-level Analysis.* At the individual level, we further explore the dynamics of program participation by investigating flow into and out of the programs. There are several mechanisms by which SSI and SNAP participation can change: a change in the rate at which individuals enter the program, both among newly eligibles and those who remain eligible but did not take up the benefit at the previous period; a change in the rate at which participants exit the program, both exiting due to loss of eligibility and exiting while still eligible. Even if entry and exit rates conditional on eligibility do not change, caseloads will change if there is an increase or decrease in the number of individuals eligible for the program.

The project decomposes each program's caseload into these specific mechanical components. The approach is similar to Cody et al. (2005), Mabli et al (2009) and Johnson (2011). The expected change in each parameter as recessions end and expansions begin is clear: fewer people should become eligible since overall incomes tend to increase with economic growth, more people exit from both eligibility and enrollment, and fewer already-eligible and newly-eligible individuals are awarded new benefits. But increasing caseloads during 2003-2007 indicate that at least one of these rates must have changed in a counter-intuitive direction.

We first conduct a descriptive analysis of the dynamics of SSI/SNAP participation to investigate the persistent growth in SSI and SNAP since 2000. We determine SNAP eligibility accounting for the gross and net income tests; the dependent, shelter, and medical expenditure deductions; and categorical eligibility. But we ignore the asset eligibility test for three reasons. First, while the SIPP has information on assets, it comes from the special topical module of the SIPP, which is only asked infrequently (the maximum is once each year per panel and varies substantially by SIPP panel). While the monthly asset information can be estimated using linear extrapolation, the method does not reflect potential fluctuations of assets, a potentially large bias given the vulnerable population we are studying. Second, the existing literature suggests that

13

asset variables in the SIPP suffer from measurement error (Strand, Rupp, and Davies 2009). Finally, the literature also suggests that income limits are much more likely to be binding than asset limits (e.g., Coe and Wu 2013). For the same reasons, we ignore the asset test in determining SSI eligibility as well. We also ignore the health requirement, as the SIPP's limited set of self-reported health measures do not correspond with the criteria used by Social Security disability examiners to decide who is medically eligible for SSI.<sup>10</sup> As a result, the fraction of the sample appearing to be eligible for SSI using our eligibility definition is larger than the proportion of that sample that would consider applying for SSI benefits.

Figure 7 shows the changes in the share of the SIPP sample eligible for each program over time. The fractions of the population eligible for SSI and SNAP remain roughly constant during the mid-2000s economic expansion at around 28 and 17 percent, respectively. It appears unlikely, therefore, that the increase in the number of SSI/SNAP eligible individuals was the cause for the unexpected increase in participation during this recovery. As the Great Recession begins, the share eligible increases dramatically, which contributes to the substantial rise in the caseload during the crisis. The aggregate pattern masks more complex trends by age group. While the pattern of eligibility among the non-elderly largely mimics the overall trend, the elderly eligibility rate decreases from 1996 to the mid-2000s, and remains constant thereafter for both SSI and SNAP.

A change in the entry rate could lead to a change in the SSI/SNAP caseload. We examine this factor by separately analyzing the change in the entry rate among newly eligibles and among those who remain eligible but did not take up the benefit in the previous interview wave. As seen in Figure 8A, the entry rate among newly eligible has increased over time for both SSI and SNAP. Rather than declining during the mid-2000s economic expansion, the entry rate for SNAP increases from about 5 percent to over 7 percent from 2003 to 2005. The entry rate also significantly increases over the Great Recession. Overall the entry rate among eligibles who did not participate in the previous period stays largely constant until the Great Recession, and rises dramatically when the recession begins (Figure 8B).

<sup>&</sup>lt;sup>10</sup> Moreover, SIPP collects these health measures infrequently, so it may miss health shocks that increase the odds of acceptance. In practice, a substantial number of SSI applicants do not report a work-limiting condition or limitation in the Activities of Daily Living in the most recent health module before they apply, even in studies that rely on administrative SSI application data (Rutledge 2012).

Another mechanism that could cause a change in SSI/SNAP participation is the change in the exit rate, both exiting due to loss of eligibility and exiting while still eligible. Figure 9A shows that the exit rate for those who remain eligible. The exit rate has decreased dramatically from 1996 to the beginning of the Great Recession, then remains relatively constant through the Great Recession. A similar pattern is observed for both the elderly and the non-elderly (not shown). The exit rate due to loss of eligibility also declines during the mid-2000 expansion (Figure 9B).<sup>11</sup>

The descriptive analysis suggests that the deceases in the exit rate among those who remain eligible and those who lose eligibility, and the increases in the entry rate among newly eligible, are the main reasons behind the increase in SSI and SNAP participation during the mid-2000s economic expansion. The increased share of the eligible and the rising entry rate also explains the substantial increase in program caseloads during the Great Recession.

To determine the underlying cause for the declining exit rate and increasing entry rate, we estimate probit regressions of the following form:

$$P_{it} = \propto Policy_{it} + X_{it}\beta + \delta_t + \varepsilon_{it} \quad (2)$$

where *P* is a binary variable for entry or exit, depending on the specification. In the estimation, *P* is modeled as a function of individuals' characteristics (*X*), including age, gender, race, marital status, education, household size, whether the individual (SSI) or household (SNAP) has kids, labor force participation status, health status, income, home/car ownership, whether *i* receives other welfare, whether *i* is living in an MSA, and state of residence. *P* is also a function of policy variables, include the state unemployment rate, minimum wage, TANF generosity, and other program specific policies, such as expanded categorical eligibility, share of SNAP recipients having short certification period, and SSI maximum benefit levels.  $\delta_t$  captures the time trends. To test whether the responsiveness has changed over time, we interact several key variables with time period dummies.

A major focus of this project is investigating the differences in entry and exit rates between the elderly and non-elderly and exploring different explanations for these transition rates for each group. Therefore, the analysis of entry and exit is estimated separately for the elderly and the non-elderly.

<sup>&</sup>lt;sup>11</sup> This part of finding is similar to Johnson (2011).

Tables 2A (SNAP) and B (SSI) summarizes the descriptive statistics for variables used in the individual-level regression. Our sample includes nearly 300,000 individuals (and close to 10million person-months) age 18 and over. Not surprisingly, SNAP and SSI eligibles are more likely to be female, minorities, low-educated, have kids, receive other welfare benefits and less likely to be married and in the labor force than those who are not eligible. On average, eligibles have lower incomes and are less likely to own a home or a car. Additionally, eligibles are in worse health. Consistent with the literature, we find the same pattern comparing eligible nonparticipants to eligible participants: in short, eligible non-participants are of higher socioeconomic status.

# 5. Results

*State-Level Results.* Table 3 presents the OLS regression results with state-fixed effects for SNAP caseload. Column (1) is the baseline regression, Column (2) – (5) allows for the effect of certain factors vary over time.

Because the main focus of the study is the counter-cyclical growth of public program caseloads over economic expansion, we interact the unemployment rate with period dummies to investigate whether the responsiveness of each public program to the business cycle has changed over time. The recovery of 2003-2007 is unique in that the poverty rate did not fall with the unemployment rate over this period. For this reason, we also interact the poverty rate with period dummies. More interestingly, the share of population with self-reported poor or fair health has increased over time, but the trends are different between the elderly and the non-elderly: the share of non-elderly in poor or fair health has trended upward, while the share of elderly with self-assessed poor health condition has declined modestly over time. Figure 10 summarizes trends for three variables used in the interaction models.

Not surprisingly, Column (1) indicates that SNAP caseload has statistically significant correlations with several measures of economic conditions, a household's demographics and health situation, and policy variables. We find that higher levels of unemployment rate and poverty rate, an increased share of the state's population in poor or fair health, and a higher out-of-pocket medical expenditure are associated with greater SNAP participation. While the literature suggests a negative age gradient in SNAP participation, the coefficient of the share of elderly population is positive, but only significant at the margin. One puzzling result is the

16

negative correlation between the fraction of low educated population and SNAP caseload, but this correlation becomes insignificant in the interaction model. Expanded categorical eligibility and the increased length of certification period also contribute to the higher caseloads, while the generosity of TANF and UI reduce SNAP caseloads.

Results from specifications that allow effects to vary over time offer several reasons that SNAP caseloads increased during the mid-2000s expansion and skyrocket during the Great Recession. First, participation becomes more cyclical with a statistically significant positive correlation between SNAP caseload and unemployment growing even stronger during the Great Recession (Column 2). Second, the positive correlation between a state's poverty rate and its SNAP enrollment first declined during 2000 to 2003, then has grown stronger thereafter (Column 3). Since the poverty rate did not fall with the unemployment rate during the economic recovery, the increased responsiveness to the poverty rate contributes to the rising caseload. Finally, as the share of the state's population reporting fair or poor health has trended upward, the positive correlation between the proportion in fair or poor health and the caseload has also grown stronger (Column 4). When all three sets of interactions are included, SNAP caseloads are less cyclical (with respect to the unemployment rate) in the 2000-2003 and 2004-2007 periods; have a positive but statistically insignificant relationship with the poverty rate (likely limited by collinearity with the unemployment rate); and a positive and statistically significant relationship with the proportion in fair or poor health (Column 5). Taken together, the greater responsiveness of the caseload to poverty levels and health, when coupled with changes in the mean values of these variables, offer an explanation for the continuing growth of the SNAP since 2000, including expansionary periods.

The aggregate pattern masks a more complex relationship by age group (Table 4). The findings for the non-elderly largely mimic those for the overall population. SNAP non-elderly caseloads were less cyclical during the 2000-2003 and 2004-2007 periods, before growing more cyclical during the Great Recession. Overall, there is no significant relationship between elderly caseloads and unemployment rate, with the exception of the Great Recession. While the non-elderly SNAP caseload is increasingly responsive to the poverty rate in the 2004-2007 period, the responsiveness of the elderly declines between 2000 and 2003, then remains constant thereafter. The correlation between elderly SNAP enrollment and the share reporting poor or fair health has become weaker over time, while the proportion of the elderly reporting fair or

17

poor health has declined. In contrast, the correlation between ill health and non-elderly SNAP caseload is strongly positive overall, and grows even stronger starting in 2004.

For the SSI program, the rich data allow us to explore both stocks (caseloads) and flows (applications and first-time awards) at the state level. Table 5 summarizes the results for SSI caseload.<sup>12</sup> While the unemployment rate is positively associated with SNAP caseload, it is insensitive to SSI caseload in the specification without interactions (Column 1). A higher share black in the state is positively associated with SSI caseload and increased fraction of the elderly reduces SSI rolls. While a greater proportion in fair or poor health and a higher out-of-pocket medical expenditure are positively associated with SSI participation, we find a negative correlation between obesity rate and SSI caseload, which is puzzling. We also find that the generosity of the TANF is negatively associated with SSI caseload and strict state regulation in the non-group health insurance market is negatively correlated with the caseload.

The interaction model shows that SSI caseloads have grown slightly more cyclical: while SSI enrollment is negatively correlated with the unemployment rate overall, this relationship has grown less negative over time and even turned positive during the Great Recession, just as unemployment rates spiked upward. Unlike SNAP, however, the relationship between overall SSI caseload and the share in poor health remains stable over time.

Table 6 examines the non-elderly and elderly separately. As with SNAP, we find opposing relationships between SSI caseloads and self-reported health condition between the elderly and the non-elderly: the positive correlation between non-elderly SSI caseloads and poor health grew stronger, and the correlation between elderly SSI caseloads and poor health became weaker and even turned negative. This finding helps to explain the unexpected caseload rise in the mid-2000s and the continuing upward trend over time.

While the unemployment rate is negatively associated with SSI caseload, it is positively correlated to application levels (Table 7) and first-time awards (Table 8). Further, having a Republican governor is correlated with lower first-time awards, while higher SSI state generosity is associated with increasing first-time awards. For SSI applications – but not awards – we find that increasing responsiveness to poor health and a weakening correlation with the

<sup>&</sup>lt;sup>12</sup> Since the poverty rate is not statistically significant in the SSI regression, we did not include it in the interaction model.

unemployment rate help explain the unexpected growth in SSI applications during the expansion and continuing upward trend since 2000.

In sensitivity checks, we also control for the lagged dependent variable – caseload applications, or first-time awards – and the results are largely similar. To pick up any discouraged worker effect, we also estimate the state regressions using the non-employment rate (100 less the labor force participation rate) in place of the unemployment rate, and the results are broadly consistent.

*Individual-Level Results.* Our individual-level analysis of unconditional trends indicates that a fall in the rate at which participants who remain eligible leave the program and the rise in the rate at which newly eligibles enter the program are the main drivers of the continuing growth in SSI and SNAP since 2000, and explain the unexpected increase in SSI and SNAP caseloads during the mid-2000s economic expansion. The regression analysis further investigates reasons behind these changes.

Table 9 summarizes the probit regression results for transitions into and out of SNAP participation. We report marginal effects – that is, the mean derivative of the outcome variable with respect to each variable – with standard errors calculated by the Delta method. Column (1) describes factors associated with exiting program while remaining eligible, Column (2) summarizes results for exit rate due to loss of eligibility, and Column (3) presents results for entry rate among newly eligibles.<sup>13</sup>

Most factors that the literature suggests should impact the public program participation decision have the expected correlation with SNAP exit among those who remain eligible (Column 1). Females, older individuals, blacks, those who have kids, and recipients of other welfare, are less likely to exit while remaining eligible, while those who are married, college-educated, higher income, currently working, living in a larger household, homeowners, and car owners are more likely to voluntarily drop out from the program. Interestingly, most of our policy variables are insignificant except share of the population with a certification period under 3 months, which is positively associated with the likelihood of exiting the program, suggesting

<sup>&</sup>lt;sup>13</sup> Most estimates are in the expected direction for two other outcomes with more stable trends in recent years: entry among those already eligible and the probability of becoming eligible.

administrative burdens deter SNAP participation.<sup>14</sup> We also find that a higher state minimum wage is negatively correlated with the exiting probability, which is bit puzzling. Further, we find that there is no correlation between self-reported health status and exiting from the program while remaining eligible.

The findings are largely similar for exiting due to losing eligibility (Column 2), except that the correlation between fair or poor health status and exiting is negative and statistically significant. Further, the correlation between working and exiting is also stronger. These findings suggest that SNAP recipients lose their eligibility primarily through moving back into the labor force or experiencing improved health.

Most variables associated with exiting are also significantly associated with entry, but in the opposite direction. While there is no correlation between health status and dropping out from the program while still eligible, the correlation is strong and positive between poor health status and entry among the newly eligible. Further, we find that a higher unemployment rate is positively associated with entry rate among the newly eligible and the likelihood of entry is lower in a state that has a large share of recipients with shorter certification period. Expanded categorical eligibility is also positively correlated with the entry decision.

The results for SNAP by age group are summarized in Table 10. The patterns between the elderly and the non-elderly are largely consistent, with a few notable exceptions. Gender, race, education, and asset ownership do not impact the exit decision among the eligible elderly, while these characteristics are significantly associated with the propensity of exiting for the nonelderly. Further, the probability that non-elderly recipients exit the program after becoming ineligible is negatively correlated with health status, while health variable is uncorrelated with exit among the elderly who remain eligible.

Table 11 summarizes the results for the SSI program. The findings are largely consistent with the SNAP analysis, except that the exiting decision among those remaining eligible is more responsive to self-assessed health status; this result is to be expected, as SSI is a disability program for the non-elderly population. While having kids is negatively associated with the likelihood of dropping out of SNAP, it is positively correlated with exiting SSI, which is

<sup>&</sup>lt;sup>14</sup> Our inclusion of state fixed effects reduces our chances of finding a statistically significant relationship between SNAP or SSI transitions and the policy variables that change very rarely, like the minimum wage and maximum TANF and SSI benefits. In the next draft of this paper, we will include results with region dummies instead of state dummies.

puzzling. The analysis by age group is largely consistent with that of the overall population (Table 12). Similar to SNAP, we find that the exiting decision among the elderly who remain eligible for SSI is unresponsive to health status.

To investigate how the correlations with each factor have changed over time, we interact several key variables with time period dummies. Tables 13A and B highlight coefficients for the main findings. Consistent with our findings in the state-level analysis, the changing responsiveness to health status, unemployment rate, and income levels are the major drivers for the changing participation in public programs over time. Further, we find that the change over time in the relationships between these factors and the two public programs differ.

For SNAP, we find individuals in poor or fair health are increasingly unlikely to exit from the program despite remaining eligible. We also find a correlation between work status and the likelihood of exiting (both those who remain eligible and those who lose eligibility) that switches signs over our sample period: while those who are working have a higher probability of exiting the program during 1996 to 1999, the likelihood of exiting among the employed declines, and then turns negative, so that the employed are less likely to exit (through either eligibility channel) over time.

With SSI, the three variables interacted with time period are less revelatory. Still, several patterns emerge. Lower-income individuals are less likely to exit SSI despite remaining eligible in all periods, and vice versa, and that relationship grows stronger in later periods. On the other hand, higher income individuals are more likely to enter SSI after becoming eligible in the 2004-2007 expansion, compared to the other three periods. Newly-eligible individuals reporting fair or poor health are more likely to enter SSI in 2004-2007 than in the other periods.

## 6. Conclusion

In tough times, SNAP and SSI help low-income households and individuals make ends meet. Recessions increase the number of people in need, so it is no surprise that SNAP and SSI rolls increase when the unemployment rate climbs. The natural expectation is that the rising tide of recovery will raise even the lowest-lying of boats, and that public program caseloads will fall during the ensuing expansion. But in the two most recent expansions – 2004-2007 and the nascent recovery from the Great Recession – SNAP and SSI participation have been stable or even increased.

21

The results of this study indicate that the cyclicality of SNAP and SSI have changed over time, and that each program is responding to different factors. States with higher unemployment rates generally see higher SNAP enrollment, but during the 2004-2007 expansion, the correlation becomes less positive. Instead, non-elderly SNAP enrollment increasingly followed the share in poverty – which did not fall during the expansion – and the share in fair or poor health – which rose throughout the period. Moreover, elderly health improved, but the relationship between elderly SNAP enrollment and health grew weaker. The individual-level results suggest that fewer SNAP beneficiaries in fair or poor health chose to leave the program during this period; more surprisingly, fewer employed people left as well.

SSI caseloads, on the other hand, historically do not increase with the unemployment rate, though applications are quite cyclical. But our state-level results suggest that SSI participation (among the non-elderly in particular) has become more responsive to the unemployment rate, even while SSI application rates have become less responsive. The individual-level analysis suggests that lower-income individuals have become less likely to leave SSI over time. Meanwhile, the state- and individual-level results indicate that people in fair or poor health have become more likely to apply to SSI and eventually enter the SSI rolls, respectively.

Though these results are informative on the cyclicality of SNAP and SSI participation, the reader should exercise caution in interpreting any results as causal. The concern is that the program entry and exit decisions are endogenous to labor market prospects, the resources available to the individual, and the individual's decision to live in a particular state. Moreover, our eligibility measure ignores the asset tests for SNAP and SSI and the medical impairment test for non-elderly SSI recipients, and thus could be subject to measurement error; in the next draft, we plan to test the robustness of our results to different eligibility criteria, including using only gross income cutoffs for SNAP (to capture measurement error in deductions) and restrictions by self-reported health measures for SSI.

We also plan to include several other robustness checks in the next draft. First, previous studies suggest that the duration of benefits influences exit probabilities (Wu 2009); to account for this, we plan to include an indicator of ongoing spell length, with separate estimates for censored and non-censored observations. In addition, we plan to re-estimate our regressions at the person-wave level to account for seam bias. Other researchers using SIPP have found that a

22

disproportionate number of program transitions occur in the interview months, i.e. the fourth reference month of each wave (Ham, Li, and Shore-Sheppard 2009). In our person-wave analysis, an individual enters the program if he participates in any month in the current wave after not participating in any month in the previous wave. Similarly, an individual exits the program if he participates in no months in the current wave after participating in at least one month in the previous wave.

The federal government spent \$626 billion on SNAP and SSI between fiscal years 2008 and 2012, compared to \$385 billion in the previous five years.<sup>15</sup> The hope after a long, deep recession and a slow start to the recovery is that more SNAP and SSI beneficiaries will move off of the rolls, with fewer new beneficiaries to replace them. The results of this study emphasize that SNAP and SSI caseload increasingly depend on poverty rates and the underlying health status of the at-risk population. Therefore, a tightening labor market is hardly sufficient for caseloads to fall.

The expansion of 2004-2007 showed that poverty rates need not decline along with unemployment rates; if growth does not help the low end of the income distribution, then poverty rates will remain elevated. Furthermore, even if a broad-based economic expansion improves both unemployment and poverty rates, health will improve much more slowly. These limitations suggest that SNAP and SSI may remain a burden on the federal budget when the Great Recession is but a distant memory.

## **References:**

- Autor, David H. and Mark G. Duggan. 2003. "The Rise in the Disability Rolls and the Decline in Unemployment." *Quarterly Journal of Economics* 118(1): 157–205.
- Beatty, Christina and Stephen Fothergill. 1996. "Labour Market Adjustment in Areas of Chronic Industrial Decline: The Case of the UK Coalfields." *Regional Studies* 30: 627–640.
- Beatty, Christina and Stephen Fothergill. 2002. "Hidden Unemployment among Men: A Case Study." *Regional Studies* 36: 811–823.

<sup>&</sup>lt;sup>15</sup> Office of Management and Budget (OMB), Historical Table 8.5, <u>http://www.whitehouse.gov/omb/budget/Historicals</u>, last accessed July 22, 2013. Note that these figures use "food and nutrition assistance" as a proxy for SNAP spending.

- Bitler, Marianne P. and Hilary W. Hoynes. 2010. "The State of the Social Safety Net in the Post-Welfare Reform Era." *Brookings Papers on Economic Activity* 2010(Fall): 71-127.
- Black, Dan, Kermit Daniel, and Seth Sanders. 2002. "The Impact of Economic Conditions on Participation in Disability Programs: Evidence from the Coal Boom and Bust." American Economic Review 92(1): 27–50
- Blank, Rebecca. 2001. "What Causes Public Assistance Caseloads to Grow?" *Journal of Human Resources* 36(1): 85-118.
- Cody, Scott, Philip Gleason, Bruce Schechter, MIki Satake, and Julie Sykes. Food Stamp Program Entry and Exit: An Analysis of Participation Trends in the 1990s. Technical Report 202, Mathematica Policy Research, Inc., 2005.
- Coe, Norma B., Kelly Haverstick, Alicia H. Munnell, and Anthony Webb. 2011. "What Explains State Variation in SSDI Application Rates?" Working Paper 2011-23. Chestnut Hill, MA: The Center for Retirement Research at Boston College.
- Coe, Norma B. and April Yanyuan Wu. 2012. "What Impact Does Social Security Have on the Use of Public Assistance Programs among the Elderly?" Manuscript.
- Coe, Norma B, Stephan Lindner, April Yanyuan Wu, and Kendrew Wong. 2013. "How Do the Disabled Cope While Waiting for SSDI?" Working Paper 2013-12. Chestnut Hill, MA: Center for Retirement Research at Boston College.
- Collins, Sarah R. 2011. "Tracking Geographical Variations in Exposure to Medical Care Economic Risk: Moving Beyond One National Estimate." Presented at: "Developing a Measure of Medical Care Economic Risk," September 8. Washington, DC: National Academy of Sciences.
- Danziger, Sandra, Mary Corcoran, Sheldon Danziger, Colleen Heflin, Ariel Kalil, Judith Levine, Daniel Rosen, Kristin Seefeldt, Kristine Siefert, and Richard Tolman. 2000. "Barriers to the Employment of Welfare Recipients." In *Prosperity for All? The Economic Boom and African Americans*, edited by Robert Cherry and William M. Rodgers III, 245-278. New York, NY: Russell Sage Foundation.
- Garrett, Bowen and Sherry Glied. 2000. "Does State AFDC Generosity Affect Child SSI Participation?" *Journal of Policy Analysis and Management* 19(2): 275-295.
- Grogger, Jeffrey. 2003. "The Effects of Time Limits, the EITC, and Other Policy Changes on Welfare Use, Work, and Income among Female-Headed Families." *The Review of Economics and Statistics* 85(2): 394-408.
- Ham, John C., Xianghong Li, and Lara Shore-Sheppard. 2009. "Seam Bias, Multiple-State, Multiple-Spell Duration Models and the Employment Dynamics of Disadvantaged

Women." Working Paper 15151. Cambridge, MA: National Bureau of Economic Research.

- Johnson, Janna. 2011. "Supplemental Nutrition Assistance Program Participation during the Economic Recovery of 2003 to 2007," *Focus*, Institute for Research on Poverty, University of Wisconsin-Madison, Spring/Summer 2012, 29(1) 9-13.
- Klerman, Jacob Alex and Steven J. Haider. 2004. "A Stock-Flow Analysis of the Welfare Caseload." *Journal of Human Resources* 39(4): 865-886.
- Leonesio, Michael V. and Linda Del Bene. 2011. "The Distribution of Annual and Long-Run US Earnings 1981-2004." *Social Security Bulletin* 71(1): 17-33.
- Lindner, Stephan. 2011. "How do Unemployment Insurance Benefits Affect the Decision to Apply for Social Security Disability Insurance?" University of Michigan.
- Mabli, James and Carolina Ferrerosa. 2010. Supplemental Nutrition Assistance Program Caseload Trends and Changes in Measures of Unemployment, Labor Underutilization, and Program Policy from 2000 to 2008. Technical report, Mathematica Policy Research, Inc., 2010.
- Mabli, James, Emily Sama Martin, and Laura Castner. Effects of Economic Conditions and Program Policy on State Food Stamp Program Caseloads, 2000 to 2006. Technical Report 56, Mathematica Policy Research, Inc., 2009.
- Maestas, Nicole, Kathleen Mullen, and Alexander Strand. 2010. "Does Disability Insurance Receipt Discourage Work? Using Examiner Assignment to Estimate Causal Effects of SSDI Receipt." Michigan Retirement Research Center working paper.
- McVicar, Duncan. 2006. "Why Do Disability Benefit Rolls Vary Between Regions? A Review of the Evidence from the USA and the UK." *Regional Studies* 40: 519–33.
- Meyer, Bruce D., Wallace K.C. Mok, and James X. Sullivan. 2009. "The Under-Reporting of Transfers in Household Surveys: Its Nature and Consequences." NBER Working Paper 15182.
- Nadel, Mark, Steve Wamhoff, and Michael Wiseman. 2003/2004. "Disability, Welfare Reform, and Supplemental Security Income." *Social Security Bulletin* 65(3): 14-30.
- Pavetti, LaDonna A. and Jacqueline Kauff. 2006. "When Five Years is not Enough: Identifying and Addressing the Needs of Families Nearing the TANF Time Limit in Ramsey County, Minnesota." Mathematica Policy Research.
- Rupp, Kalman and David Stapleton. 1995. "Determinants of the Growth in the Social Security Administration's Disability Programs." *Social Security Bulletin* 58(4): 43-70.

- Rutledge, Mathew S. 2012. "The Impact of Unemployment Insurance Extensions on Disability Insurance Application and Allowance Rates." Working Paper 2011-17. Chestnut Hill, MA: Center for Retirement Research at Boston College.
- Schmidt, Lucie. 2012. "The Supplemental Security Income Program and Welfare Reform." Working Paper.
- Schmidt, Lucie and Purvi Sevak. 2004. "AFDC, SSI, and Welfare Reform Aggressiveness: Caseload Reductions vs. Caseload Shifting." *Journal of Human Resources* 39(3): 792-812.
- Smith, Sheila, Joseph P. Newhouse, and Mark S. Freeland. 2009. "Income, Insurance, and Technology: Why Does Health Spending Outpace Economic Growth?" *Health Affairs* 28(5): 1276-1284.
- Stapleton, David C., Kevin A. Coleman, and Kimberly A. Dietrich. 1995. "Demographic and Economic Determinants of Recent Application and Award Growth for SSA's Disability Programs." Presented at: "The Social Security Administration's Disability Programs: Explanations of Recent Growth and Implications for Disability Policy." Washington, DC: Social Security Administration and the U.S. Department of Health and Human Services.
- Stapleton, David C., Kevin A. Coleman, Kimberly A. Dietrich, and Gina A. Livermore. 1998. "Empirical Analysis of DI and SSI Application and Award Growth." In *Growth in Disability Benefits: Explanations and Policy Implications*, edited by Kalman Rupp and David C. Stapleton, 31-92. Kalamazoo, MI: W.E. Upjohn Institute for Employment Research.
- Stapleton, David C., Michael Fishman, Gina A. Livermore, David Wittenburg, Adam Tucker, and Scott Scrivner. 1999. Policy Evaluation of the Overall Effects of Welfare Reform on SSA Programs: Final Report. Falls Church, VA: The Lewin Group (for the Social Security Administration).
- Strand, Alexander, Kalman Rupp and Paul S. Davies. 2009. "Measurement Error in Estimates of the Participation Rate in Means-Tested Programs: The Case of the US Supplemental Security Income Program for the Elderly." Working Paper.

Social Security Administration. Statistics Annual Supplement, 1990-2011. Washington, DC.

- Wolkwitz, Kari, and Carole Trippe. —Characteristics of Supplemental Nutrition Assistance Program Households: Fiscal Year 2008. Alexandria, VA: Food and Nutrition Service, U.S. Department of Agriculture, September 2009.
- Wu, April Yanyuan. 2009. "Why Do So Few Elderly Use Food Stamps?" Working Paper 10-01. Chicago, IL: The Harris School of Public Policy Studies, University of Chicago.

Ziliak, James P., Craig Gundersen, and David N. Figlio. 2003. "Food Stamp Caseloads over the Business Cycle." *Southern Economic Journal* 69(4): 903-919.

Figure 1: SNAP Caseload and SSI Applications and Awards



Source: U.S. Social Security Administration and U.S. Department of Agriculture. 1982-2011. Washington, DC.

Figure 2: SNAP Caseload and SSI Applications, by Age Group



Source: U.S. Social Security Administration and U.S. Department of Agriculture. 1982-2011. Washington, DC.



Figure 3: Percentage Point Change in SSI Caseload/State Population; 1996-2011

Source: Authors' Calculations

Figure 4: Percentage Point Change in SSI Application Rates; 1996-2011



Source: Authors' Calculations



Figure 5: Percentage Point Change in SSI 1st Time Awards/State Population; 1996-2011

Source: Authors' Calculations

Figure 6: Percentage Point Change in SNAP Caseload/State Population; 1996-2011



Source: Authors' Calculations



Figure 7A. Proportion Eligible for SNAP and SSI, Age 18+

Source: Survey of Income and Program Participation, 1996-2008 Panels.

Figure 7B. Proportion Eligible for SNAP and SSI, Non-Elderly



Notes: Sample includes all individuals under 60 for SNAP and under 65 for SSI. *Source*: Survey of Income and Program Participation, 1996-2008 Panels.



Figure 7C. Proportion Eligible for SNAP and SSI, Elderly

Notes: Sample includes all individuals 60 and over for SNAP and 65 and over for SSI. *Source*: Survey of Income and Program Participation, 1996-2008 Panels.

Figure 8A. Entry Rate for Newly Eligibles



Notes: Sample includes individuals who are ineligible in (t-1), and are eligible in (t). The numerator is individuals observed participating in (t).

Source: Survey of Income and Program Participation, 1996-2008 Panels.



Figure 8B. Entry Rate for Already-Eligible Non-Participants

Notes: Sample includes individuals who are eligible for both (t-1) and (t), and don't participate in (t-1). The numerator is individuals observed participating in (t).

Source: Survey of Income and Program Participation, 1996-2008 Panels.



Figure 9A. Exit Rate for Beneficiaries Who Remain Eligible

Notes: Sample includes individuals who are eligible for both (t-1) and (t), and participate in (t-1). The numerator is individuals observed not participating in (t).

Source: Survey of Income and Program Participation, 1996-2008 Panels.



Figure 9B. Exit Rate for Beneficiaries Who Lose Eligibility

Notes: Sample includes individuals who are eligible and participate in (t-1). The numerator is individuals observed becoming ineligible in (t).

Source: Survey of Income and Program Participation, 1996-2008 Panels.



Figure 10: Unemployment, Poverty, and Self-Reported General Health; 1996-2011

					Within-
		Between St	ates Over Time	<b>`</b>	Time
		Standard			Standard
	Mean	Deviation	Minimum	Maximum	Deviation
Dependent Variables					
Total SNAP Caseload	0.101	0.044	0.035	0.279	0.032
Elderly SNAP Caseload (60+)	0.057	0.025	0.005	0.182	0.013
Non-Elderly SNAP Caseload	0.122	0.058	0.038	0.375	0.045
Total SSI Caseload	0.030	0.012	0.013	0.074	0.002
Elderly SSI Caseload (65+)	0.045	0.026	0.010	0.165	0.008
Non-Elderly SSI Caseload	0.021	0.009	0.009	0.053	0.001
Total New SSI Awards	0.004	0.001	0.002	0.008	0.000
Total New Elderly SSI Awards	0.002	0.002	0.000	0.017	0.001
Total New Non-Elderly SSI Awards	0.003	0.001	0.001	0.007	0.000
SSI Application Rate	0.008	0.003	0.001	0.027	0.002
Economic Variables					
Unemployment Rate	0.057	0.022	0.014	0.150	0.019
Change in 10th Percentile of 3 Year Moving					
Average of Real Wage Distribution	0.007	0.021	-0.054	0.095	0.021
Poverty Rate	0.124	0.034	0.045	0.255	0.017
Minimum Wage	5.793	1.031	4.250	8.670	0.934
Service Occupation	0.480	0.088	0.307	0.771	0.076
Manufacturing Job	0.126	0.054	0.012	0.278	0.025
Health Care Expenditure (2011 Dollars)	6307 449	1291 552	3593 395	11169 972	954 420
Demographics and Health Variables	00071115	12)1.002	0070.070	11107.572	<i>y</i> <b>c2</b> 0
Fraction Black	0 1 1 4	0 120	0.000	0 711	0.012
Fraction Newly Arrived Immigrants	0.013	0.009	0.000	0.047	0.005
Fraction elderly (60+)	0.168	0.025	0.057	0.236	0.014
Fraction Male	0.483	0.023	0.448	0.520	0.007
Fraction Single Mother Household	0.332	0.061	0.184	0.520	0.007
Fraction Less than High School	0.139	0.001	0.060	0.257	0.020
Fraction Obese 18+	0.227	0.047	0.000	0.356	0.039
Fraction Obese 18-59	0.229	0.050	0.097	0.374	0.041
Fraction Obese 60+	0.222	0.030	0.050	0.374	0.038
Fraction Reporting Fair or Poor Health 18+	0.150	0.045	0.082	0.254	0.014
Fraction Reporting Fair or Poor Health 18-59	0.114	0.034	0.053	0.234	0.014
Fraction Reporting Fair or Poor Health 60+	0.114	0.055	0.028	0.212	0.023
Policy Variables	0.200	0.055	0.020	0.404	0.025
Categorical SNAP eligibility	0 221	0.415	0.000	1 000	0 341
Eraction Short SNAP Recertification Period	0.084	0.130	0.000	0.675	0.105
SNAD Error Date	6 877	2 1 / 2	0.000	17 670	2 820
Maximum State Supplement to SSI	38.025	70 757	0.700	520.270	2.829
Strict Degulation	0.124	0.330	0.000	1 000	0.125
Republican Governor	0.124	0.550	0.000	1.000	0.125
May TANE benefits	0.545 A16 152	0.470	120.000	072 000	20 272
Madicaid Ruy In	410.133	0 500	0.000	1 000	0.72
Maximum Waakhy III Danafit	0.499	0.300 80.025	151.000	620.000	0.421
waxiiiuiii weekiy 01 Dellelli	555.002	09.923	131.000	029.000	50.925

Source: Authors' Calculations

	SNAP	SNAP	SNAP Eligible	SNAP Eligible
	Ineligible	Eligible	Nonparticipants	Participants
Age	45.6	45.4	46.0	43.3
Female	0.505	0.586	0.538	0.767
Marital Status	0.602	0.386	0.430	0.218
Race				
White	0.846	0.723	0.754	0.604
Black	0.098	0.204	0.172	0.327
Other	0.056	0.073	0.074	0.069
Education				
Less than HS	0.095	0.292	0.275	0.358
HS	0.293	0.346	0.343	0.354
College	0.612	0.362	0.382	0.288
Household Size	3.0	3.1	3.0	3.1
Kids	0.369	0.439	0.408	0.559
Live in an MSA	0.799	0.755	0.758	0.742
Employment	0.712	0.367	0.396	0.254
Poor or Fair Health	0.122	0.306	0.277	0.416
Income (monthly)				
Mean	\$3400	\$700	\$700	\$500
Median	\$2500	\$400	\$500	\$300
Own Home	0.765	0.471	0.528	0.257
Own a Vehicle	0.905	0.683	0.730	0.504
Receive Other				
Welfare	0.115	0.527	0.428	0.902
N	8,020,423	1,873,880	1,455,902	417,978

Table 2A. Summary Statistics for SNAP

Source: Authors' Calculations

	SSI	SSI	SSI Eligible	SSI Eligible
	Ineligible	Eligible	Nonparticipants	Participants
Age	49.4	36.7	35.4	51.8
Female	0.488	0.592	0.591	0.603
Marital Status	0.639	0.385	0.400	0.208
Race				
White	0.852	0.759	0.771	0.625
Black	0.098	0.161	0.151	0.276
Other	0.050	0.079	0.078	0.099
Education				
Less than HS	0.094	0.217	0.196	0.458
HS	0.284	0.347	0.349	0.330
College	0.622	0.435	0.455	0.212
Household Size	2.8	3.5	3.6	2.7
Kids	0.337	0.485	0.506	0.241
Live in an MSA	0.795	0.781	0.784	0.750
Employment	0.728	0.467	0.502	0.067
Poor or Fair Health	0.144	0.182	0.143	0.620
Income (monthly)				
Mean	\$3900	\$500	\$600	\$200
Median	\$2900	\$200	\$300	\$0
Own Home	0.773	0.569	0.585	0.389
Own a Vehicle	0.898	0.787	0.812	0.505
Receive Other				
Welfare	0.111	0.393	0.342	0.981
Ν	6,937,911	2,956,392	2,692,831	263,561

Table 2B. Summary Statistics for SSI

Source: Authors' Calculations

SNAP Caseload			18+		
	(1)	(2)	(3)	(4)	(5)
Economic Variables					
Unemployment Rate	0.5486***	0.3764***	0.5131***	0.5365***	0.4585***
	(0.0418)	(0.0920)	(0.0408)	(0.0403)	(0.0973)
Change in 10th Percentile of 3 Year	0.0085	-0.0220	-0.0052	-0.0019	-0.0240
Moving Average of Real Wage Distribution	(0.0222)	(0.0245)	(0.0221)	(0.0214)	(0.0234)
Poverty Rate	0.1482***	0.1327***	0.0908*	0.1333***	0.1005**
	(0.0452)	(0.0399)	(0.0462)	(0.0429)	(0.0428)
Health Care Expenditure (\$1000 of	0.0090***	0.0103***	0.0109***	0.0118***	0.0123***
2011)	(0.0027)	(0.0034)	(0.0024)	(0.0028)	(0.0031)
Service Occupation	-0.0039	0.0193	-0.0015	0.0006	0.0202
	(0.0167)	(0.0156)	(0.0164)	(0.0169)	(0.0155)
Manufacturing Job	-0.0620	-0.0694*	-0.0763**	-0.0489	-0.0632*
	(0.0414)	(0.0348)	(0.0361)	(0.0383)	(0.0341)
Demographics and Health Variables					
Fraction Black	-0.0000	0.0100	0.0612	0.0128	0.0377
	(0.0514)	(0.0446)	(0.0429)	(0.0605)	(0.0488)
Fraction Newly Arrived Immigrants	-0.1055	-0.0513	-0.1258	-0.1365	-0.0715
	(0.1441)	(0.1357)	(0.1359)	(0.1353)	(0.1342)
Fraction 60+	0.1319*	0.0761	0.1259*	0.1156	0.0818
	(0.0746)	(0.0673)	(0.0728)	(0.0696)	(0.0661)
Fraction Male	0.0075	-0.0185	-0.0137	-0.0800	-0.0788
	(0.0877)	(0.0827)	(0.0843)	(0.0889)	(0.0856)
Fraction Single Mother Household	-0.0170	-0.0105	-0.0263	-0.0385	-0.0287
	(0.0367)	(0.0349)	(0.0359)	(0.0372)	(0.0354)
Fraction Less Than High School	-0.1108**	-0.0477	-0.0650	-0.0651	-0.0184
	(0.0504)	(0.0473)	(0.0509)	(0.0462)	(0.0475)
Fraction Obese by Age Group	-0.0499	-0.0350	-0.0743**	-0.0783**	-0.0728**
	(0.0366)	(0.0367)	(0.0361)	(0.0369)	(0.0359)
Fraction Reporting Fair or Poor	0.2138***	0.2143***	0.2066***	0.1232**	0.1408**
Health by Age Group	(0.0480)	(0.0500)	(0.0446)	(0.0553)	(0.0586)
Policy Variables	~ /	· · · ·		× /	
Categorical SNAP eligibility	0.0122***	0.0110***	0.0115***	0.0121***	0.0111***
	(0.0031)	(0.0030)	(0.0031)	(0.0031)	(0.0030)
Fraction with Short SNAP	-0.0378***	-0.0423***	-0.0350***	-0.0329***	-0.0360***
Recertification Period	(0.0077)	(0.0071)	(0.0065)	(0.0066)	(0.0061)
SNAP Combined Error Rate	0.0001	0.0000	-0.0002	-0.0001	-0.0001
	(0.0003)	(0.0003)	(0.0003)	(0.0003)	(0.0003)
Strict Regulation	-0.0105	-0.0103	-0.0124*	-0.0098	-0.0102
-	(0.0080)	(0.0072)	(0.0063)	(0.0069)	(0.0062)
Republican Governor	-0.0008	-0.0010	-0.0011	-0.0010	-0.0011
-	(0.0019)	(0.0018)	(0.0017)	(0.0018)	(0.0017)

Table 3. State-Level Regressions: SNAP Caseload 18+

1 able 3. State-Level Regressions: SNAP Caseload 18+ (con	Table 3.	State-Level	Regressions:	<b>SNAP</b>	<i>Caseload</i> 18+	(cont'd	1)
---	----------	-------------	--------------	-------------	---------------------	---------	----

AP Caseload			18+		
	(1)	(2)	(3)	(4)	(5)
Max TANF benefits (Hundreds	-0.0042*	-0.0030	-0.0034	-0.0043*	-0.0033
of Dollars)	(0.0025)	(0.0024)	(0.0021)	(0.0024)	(0.0022)
Medicaid Buy-In	0.0002	0.0010	-0.0009	-0.0005	0.0003
	(0.0023)	(0.0022)	(0.0023)	(0.0023)	(0.0023)
Maximum Weekly UI Benefit	-0.0150***	-0.0136***	-0.0125***	-0.0137***	-0.0125***
(\$100)	(0.0030)	(0.0027)	(0.0028)	(0.0030)	(0.0026)
Minimum Wage	0.0018	-0.0004	0.0009	0.0014	0.0000
	(0.0012)	(0.0013)	(0.0012)	(0.0013)	(0.0015)
2000-2003	-0.0119***	0.0031	-0.0032	-0.0074	0.0020
	(0.0024)	(0.0045)	(0.0056)	(0.0075)	(0.0074)
2004-2007	-0.0019	-0.0026	-0.0163**	-0.0262**	-0.0219*
	(0.0039)	(0.0078)	(0.0078)	(0.0100)	(0.0114)
2008-2011	0.0159***	-0.0094	-0.0155	-0.0182	-0.0335**
	(0.0047)	(0.0076)	(0.0094)	(0.0113)	(0.0127)
2000-2003*Unemployment	· · · ·	-0.3378***	( )	( )	-0.3213***
		(0.1024)			(0.1198)
2004-2007*Unemployment		-0.1024			-0.2343*
		(0.1204)			(0.1197)
2008-2011*Unemployment		0.3480***			0.2042
		(0.1192)			(0.1250)
2000-2003*Poverty		× ,	-0.0872**		-0.0422
-			(0.0412)		(0.0614)
2004-2007*Poverty			0.0986*		0.0291
-			(0.0566)		(0.0605)
2008-2011*Poverty			0.2287***		0.1087
-			(0.0633)		(0.0719)
2000-2003*% Reporting Fair or			( )	-0.0273	0.0370
Poor Health				(0.0494)	(0.0565)
2004-2007*% Reporting Fair or				0.1574**	0.1512**
Poor Health				(0.0649)	(0.0681)
2004-2007*% Reporting Fair or				0 2261***	0.1246
Poor Health				(0.0674)	(0.0760)
Constant	0 0249	0.0283	0.0230	0.0691	0.0543
Constant	(0.024)	(0.0283)	(0.0230)	(0.00)	(0.0343)
	(0.0502)	(0.0479)	(0.0440)	(0.0500)	(0.0403)
State Dummies	Yes	Yes	Yes	Yes	Yes
Observations	816	816	816	816	816
R-squared	0.9142	0.9221	0.9199	0.9192	0.9256
Number of Clusters	51	51	51	51	51

SNAP Caseload	18	-60		60+		
	(1)	(2)	(3)	(4)		
Unemployment Rate	0.8190***	0.6835***	0.0816*	-0.0363		
	(0.0589)	(0.1247)	(0.0412)	(0.0703)		
Poverty Rate	0.1699**	0.0816	0.0547	0.0983***		
	(0.0788)	(0.0681)	(0.0383)	(0.0327)		
Fraction Reporting Fair or	0.3315***	0.1062	0.0537**	0.0989***		
Poor Health by Age Group	(0.0754)	(0.0878)	(0.0227)	(0.0345)		
2000-2003	-0.0168***	-0.0048	-0.0069***	0.0240**		
	(0.0035)	(0.0081)	(0.0019)	(0.0097)		
2004-2007	-0.0050	-0.0422***	-0.0052*	0.0229		
	(0.0056)	(0.0136)	(0.0030)	(0.0147)		
2008-2011	0.0221***	-0.0646***	-0.0004	0.0234		
	(0.0067)	(0.0194)	(0.0037)	(0.0147)		
2000-2003*Unemployment		-0.4810***		-0.0620		
		(0.1599)		(0.0931)		
2004-2007*Unemployment		-0.3154*		0.0291		
		(0.1599)		(0.0830)		
2008-2011*Unemployment		0.2957*		0.1806**		
		(0.1675)		(0.0876)		
2000-2003*Poverty		-0.0248		-0.0835*		
		(0.0796)		(0.0471)		
2004-2007*Poverty		0.1207*		-0.0481		
		(0.0703)		(0.0612)		
2008-2011*Poverty		0.1499		-0.0076		
		(0.1110)		(0.0731)		
2000-2003*% Reporting Fair		0.1201		-0.0667*		
or Poor Health		(0.0832)		(0.0340)		
2004-2007*% Reporting Fair		0.2718***		-0.0932*		
or Poor Health		(0.0929)		(0.0469)		
2004-2007*% Reporting Fair		0.3688**		-0.1323***		
or Poor Health		(0.1473)		(0.0487)		
Observations	816	816	816	816		
R-squared	0.9038	0.9204	0.8404	0.8533		

Table 4. State-Level Regressions: SNAP Caseload by Age Group

SSI Caseload	Full					
	(1)	(2)	(3)	(4)		
Economic Variables						
Unemployment Rate	-0.0049	-0.0347*	-0.0054	-0.0416***		
	(0.0057)	(0.0178)	(0.0058)	(0.0150)		
Change in 10th Percentile of 3	-0.0019	-0.0040*	-0.0024	-0.0045**		
Year Moving Average of Real Wage Distribution	(0.0020)	(0.0022)	(0.0019)	(0.0021)		
Poverty Rate	-0.0079	-0.0081	-0.0072	-0.0066		
	(0.0066)	(0.0067)	(0.0059)	(0.0057)		
Health Care Expenditure (\$1000	0.0014***	0.0014***	0.0014***	0.0013***		
of 2011)	(0.0004)	(0.0004)	(0.0005)	(0.0004)		
Service Occupation	-0.0012	-0.0001	-0.0013	-0.0003		
	(0.0022)	(0.0022)	(0.0021)	(0.0022)		
Manufacturing Job	0.0078	0.0069	0.0079	0.0065		
	(0.0077)	(0.0075)	(0.0075)	(0.0069)		
Demographics and Health Variables						
Fraction Black	0.0145*	0.0173**	0.0143*	0.0176**		
	(0.0074)	(0.0070)	(0.0072)	(0.0069)		
Fraction Newly Arrived	-0.0181	-0.0135	-0.0196	-0.0139		
Immigrants	(0.0165)	(0.0159)	(0.0167)	(0.0162)		
Fraction 65+	-0.0217**	-0.0255**	-0.0222**	-0.0261**		
	(0.0102)	(0.0099)	(0.0101)	(0.0100)		
Fraction Male	-0.0227	-0.0247	-0.0218	-0.0212		
	(0.0170)	(0.0164)	(0.0133)	(0.0133)		
Fraction Single Mother	-0.0003	-0.0005	-0.0000	0.0003		
Household	(0.0033)	(0.0030)	(0.0033)	(0.0031)		
Fraction Less than High School	0.0106	0.0143*	0.0101	0.0123		
	(0.0081)	(0.0072)	(0.0090)	(0.0085)		
Fraction Obese	-0.0294***	-0.0273***	-0.0290***	-0.0255***		
	(0.0082)	(0.0088)	(0.0070)	(0.0071)		
Fraction Reporting Fair or Poor	0.0161**	0.0154*	0.0209	0.0242		
Health	(0.0074)	(0.0077)	(0.0162)	(0.0155)		

Table 5. State-Level Regressions: SSI Caseload 18+

SSI Caseload	Full				
	(1)	(2)	(3)	(4)	
Policy Variables	· · ·				
Maximum State Supplement to	0.0004	0.0001	0.0005	0.0004	
SSI (Hundreds)	(0.0010)	(0.0010)	(0.0011)	(0.0011)	
Strict Regulation	-0.0010**	-0.0010*	-0.0011**	-0.0011**	
	(0.0005)	(0.0005)	(0.0005)	(0.0005)	
Republican Governor	-0.0002	-0.0002	-0.0002	-0.0002	
	(0.0002)	(0.0002)	(0.0002)	(0.0002)	
Max TANF benefits (Hundreds)	-0.0007*	-0.0006*	-0.0006	-0.0006	
	(0.0004)	(0.0004)	(0.0004)	(0.0004)	
Medicaid Buy-In	0.0003	0.0004	0.0003	0.0003	
	(0.0003)	(0.0003)	(0.0003)	(0.0003)	
Maximum Weekly UI Benefits	0.0001	0.0003	0.0001	0.0002	
(Hundreds)	(0.0005)	(0.0005)	(0.0005)	(0.0004)	
Minimum Wage	0.0004**	0.0002	0.0004*	0.0002	
	(0.0002)	(0.0002)	(0.0002)	(0.0002)	
2000-2003	-0.0012***	-0.0016**	0.0002	-0.0002	
	(0.0002)	(0.0008)	(0.0016)	(0.0017)	
2004-2007	-0.0012***	-0.0027**	0.0000	-0.0011	
	(0.0004)	(0.0012)	(0.0025)	(0.0027)	
2008-2011	0.0000	-0.0024**	0.0005	-0.0011	
	(0.0005)	(0.0011)	(0.0027)	(0.0028)	
2000-2003*Unemployment		0.0078		0.0160	
		(0.0168)		(0.0151)	
2004-2007*Unemployment		0.0282		0.0380**	
		(0.0238)		(0.0185)	
2008-2011*Unemployment		0.0429**		0.0506***	
		(0.0192)		(0.0168)	
2000-2003*% in Fair or Poor			-0.0094	-0.0123	
Health			(0.0118)	(0.0103)	
2004-2007*% in Fair or Poor			-0.0083	-0.0146	
Health			(0.0165)	(0.0147)	
2008-2011*% in Fair or Poor			-0.0034	-0.0115	
Health			(0.0180)	(0.0172)	
Constant	0.0535***	0.0549***	0.0524***	0.0528***	
	(0.0084)	(0.0080)	(0.0066)	(0.0066)	
State Dummies	Yes	Yes	Yes	Yes	
Observations	816	816	816	816	
R-squared	0.9870	0.9875	0.9871	0.9877	
Number of Clusters	51	51	51	51	

Table 5. State-Level Regressions: SSI Caseload 18+ (cont'd)

SSI Caseload	18-	-64	65+	
	(1)	(2)	(1)	(2)
Unemployment Rate	0.0002	-0.0236**	-0.0233	-0.0366
	(0.0043)	(0.0109)	(0.0240)	(0.0593)
Poverty Rate	-0.0061	-0.0078**	-0.0263	0.0082
	(0.0037)	(0.0036)	(0.0373)	(0.0248)
Fraction Reporting Fair or	0.0112**	-0.0048	0.0313*	0.0792***
Poor Health	(0.0046)	(0.0075)	(0.0178)	(0.0258)
2000-2003	-0.0005***	-0.0024***	-0.0041***	0.0155**
	(0.0002)	(0.0008)	(0.0014)	(0.0067)
2004-2007	-0.0009***	-0.0041***	-0.0057***	0.0279**
	(0.0003)	(0.0012)	(0.0021)	(0.0114)
2008-2011	-0.0001	-0.0047***	-0.0073***	0.0425***
	(0.0004)	(0.0014)	(0.0026)	(0.0140)
2000-2003*Unemployment		0.0064		-0.0081
		(0.0128)		(0.0542)
2004-2007*Unemployment		0.0266**		0.0441
		(0.0126)		(0.0715)
2008-2011*Unemployment		0.0323**		0.0110
		(0.0126)		(0.0593)
2000-2003*% in Fair or Poor		0.0142**		-0.0660***
Health		(0.0061)		(0.0228)
2004-2007*% in Fair or Poor		0.0158**		-0.1231***
Health		(0.0077)		(0.0314)
2008-2011*% in Fair or Poor		0.0236**		-0.1824***
Health		(0.0096)		(0.0452)
Observations	816	816	816	816
R-squared	0.9875	0.9892	0.9426	0.9557

Table 6. State-Level Regressions: SSI Caseload by Age Group

SSI Applications	18+					
	(1)	(2)	(3)	(4)		
Economic Variables						
Unemployment Rate	0.0271***	0.0240***	0.0255***	0.0379***		
	(0.0041)	(0.0071)	(0.0039)	(0.0067)		
Change in 10th Percentile of 3	-0.0008	-0.0006	-0.0007	0.0001		
Year Moving Average of Real Wage Distribution	(0.0026)	(0.0026)	(0.0025)	(0.0023)		
Poverty Rate	0.0107**	0.0113**	0.0079**	0.0080**		
	(0.0041)	(0.0042)	(0.0039)	(0.0039)		
Health Care Expenditure (\$1000	0.0009***	0.0009***	0.0012***	0.0012***		
of 2011)	(0.0003)	(0.0003)	(0.0003)	(0.0003)		
Service Occupation	0.0008	0.0004	0.0006	0.0000		
	(0.0016)	(0.0015)	(0.0015)	(0.0013)		
Manufacturing Job	-0.0060	-0.0058	-0.0049	-0.0042		
	(0.0043)	(0.0045)	(0.0038)	(0.0041)		
Demographics and Health Variables						
Fraction Black	-0.0038	-0.0033	-0.0036	-0.0045		
	(0.0030)	(0.0032)	(0.0034)	(0.0040)		
Fraction Newly Arrived	0.0031	0.0036	-0.0004	-0.0025		
Immigrants	(0.0102)	(0.0102)	(0.0109)	(0.0103)		
Fraction 65+	-0.0032	-0.0028	-0.0036	-0.0019		
	(0.0049)	(0.0049)	(0.0047)	(0.0048)		
Fraction Male	0.0038	0.0032	-0.0047	-0.0054		
	(0.0059)	(0.0058)	(0.0062)	(0.0063)		
Fraction Single Mother	0.0041*	0.0040*	0.0024	0.0022		
Household	(0.0021)	(0.0020)	(0.0020)	(0.0020)		
Fraction Less than High School	-0.0094**	-0.0102**	-0.0022	-0.0035		
	(0.0042)	(0.0042)	(0.0038)	(0.0040)		
Fraction Obese	-0.0074*	-0.0083**	-0.0111***	-0.0130**		
	(0.0040)	(0.0040)	(0.0039)	(0.0039)		
Fraction Reporting Fair or Poor	0.0296***	0.0294***	0.0124**	0.0114**		
Health	(0.0056)	(0.0056)	(0.0055)	(0.0054)		

Table 7. State-Level Regressions: SSI Applications 18+

SSI Applications	18+					
	(1)	(2)	(3)	(4)		
Policy Variables						
Maximum State Supplement to	0.0002	0.0002	-0.0004	-0.0004		
SSI (Hundreds)	(0.0005)	(0.0005)	(0.0006)	(0.0006)		
Strict Regulation	-0.0003	-0.0003	-0.0001	-0.0001		
	(0.0003)	(0.0003)	(0.0003)	(0.0003)		
Republican Governor	-0.0001	-0.0001	-0.0001	-0.0001		
	(0.0001)	(0.0001)	(0.0001)	(0.0001)		
Max TANF benefits (Hundreds)	-0.0003	-0.0003	-0.0003	-0.0003		
	(0.0003)	(0.0003)	(0.0003)	(0.0003)		
Medicaid Buy-In	0.0001	0.0001	0.0001	0.0001		
	(0.0002)	(0.0002)	(0.0002)	(0.0002)		
Maximum Weekly UI Benefits	-0.0003	-0.0003	-0.0000	-0.0001		
(Hundreds)	(0.0002)	(0.0003)	(0.0003)	(0.0003)		
Minimum Wage	-0.0004***	-0.0004***	-0.0004***	-0.0003**		
	(0.0001)	(0.0001)	(0.0001)	(0.0001)		
2000-2003	0 0000	-0.0006**	-0.0023***	-0.0023***		
	(0.0002)	(0.0003)	(0.0006)	(0.0006)		
2004-2007	-0.0013***	-0.0009	-0.0035***	-0.0029***		
	(0.0003)	(0.0006)	(0.0007)	(0.0008)		
2008-2011	-0.0007**	-0.0006	-0.0059***	-0.0052***		
	(0.0003)	(0.0005)	(0.0011)	(0.0011)		
2000-2003*Unemployment	()	0.0146***	()	0.0018		
1.5		(0.0047)		(0.0064)		
2004-2007*Unemployment		-0.0044		-0.0167*		
		(0.0093)		(0.0088)		
2008-2011*Unemployment		0.0028		-0.0182**		
		(0.0089)		(0.0081)		
2000-2003*% in Fair or Poor			0.0165***	0.0167***		
Health			(0.0041)	(0.0043)		
2004-2007*% in Fair or Poor			0.0154***	0.0180***		
Health			(0.0044)	(0.0044)		
2008-2011*% in Fair or Poor			0.0358***	0.0387***		
Health			(0.0071)	(0.0069)		
Constant	0.0053	0.0062	0.0104***	0.0107***		
	(0.0038)	(0.0038)	(0.0038)	(0.0039)		
State Dummies	Yes	Yes	Yes	Yes		
Observations	816	816	816	816		
R-squared	0.8865	0.8872	0.8990	0.9003		
Number of Clusters	51	51	51	51		

 Table 7. State-Level Regressions: SSI Applications 18+ (cont'd)

SSI 1st Awards	1	8+	18	-64	6	55+
	(1)	(2)	(1)	(2)	(1)	(2)
Economic Variables						
Unemployment Rate	0.0064***	0.0010	0.0067***	0.0032	0.0009	0.0018
	(0.0011)	(0.0030)	(0.0012)	(0.0028)	(0.0015)	(0.0039)
Change in 10th Percentile of 3	-0.0009*	-0.0011**	-0.0005	-0.0008*	-0.0009	-0.0004
Year Moving Average of Real						
Wage Distribution	(0.0004)	(0.0005)	(0.0004)	(0.0004)	(0.0009)	(0.0011)
Poverty Rate	-0.0015	-0.0013	-0.0010	-0.0009	-0.0033*	-0.0029
	(0.0013)	(0.0012)	(0.0012)	(0.0011)	(0.0019)	(0.0018)
Health Care Expenditure	0.0000	0.0000	0.0001	0.0001	-0.0002	-0.0003
(\$1000 of 2011)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0002)	(0.0002)
Service Occupation	-0.0005	-0.0005	-0.0008*	-0.0006	0.0009	0.0004
	(0.0005)	(0.0005)	(0.0005)	(0.0004)	(0.0018)	(0.0013)
Manufacturing Job	-0.0007	-0.0008	-0.0016	-0.0016	0.0045*	0.0047**
	(0.0018)	(0.0018)	(0.0015)	(0.0015)	(0.0023)	(0.0023)
Demographics and Health Variable	es					
Fraction Black	-0.0038*	-0.0032	-0.0046**	-0.0043**	0.0028	0.0029
	(0.0020)	(0.0019)	(0.0020)	(0.0020)	(0.0029)	(0.0028)
Fraction Newly Arrived	0.0037	0.0046	0.0041	0.0052	-0.0045	-0.0050
Immigrants	(0.0033)	(0.0035)	(0.0036)	(0.0037)	(0.0047)	(0.0050)
Fraction 65+	-0.0003	-0.0008			· · · · ·	× ,
	(0.0018)	(0.0018)				
Fraction Male	-0.0035*	-0.0040*	-0.0033	-0.0039*	-0.0021	-0.0024
	(0.0020)	(0.0020)	(0.0020)	(0.0020)	(0.0032)	(0.0032)
Fraction Single Mother	0.0009	0.0008	0.0012*	0.0012**	-0.0011	-0.0012
Household	(0.0007)	(0.0007)	(0,0006)	(0,0006)	(0.0015)	(0.0013)
Fraction Less Than High	-0.0016	-0.0013	-0.0017	-0.0013	0.0014	0.0006
School	(0.0017)	(0.0017)	(0.0014)	(0.0013)	(0.0025)	(0.0022)
Fraction Obese	-0.0001	-0.0001	0.0003	0.0003	-0.0012	-0.0017
	(0.00018)	(0.0018)	(0,0014)	(0.0014)	(0.0012)	(0.0013)
Fraction Reporting Fair or	0.0020	0.0018	0.0074	0.0023	0.0012)	0.0020**
Poor Health	(0.0020)	(0.0010)	(0.0027)	(0.0025)	(0,00020)	(0,00020)
	(0.0020)	(0.0020)	(0.0013)	(0.0015)	(0.0009)	(0.0009)

Table 8. State-Level Regressions: SSI 1st Awards by Age Group

SSI 1st Awards		18+	18-	-64	(	65+
	(1)	(2)	(1)	(2)	(1)	(2)
Policy Variables		· ·	· · ·			
Maximum State Supplement to	0.0005***	0.0005***	0.0006***	0.0006***	0.0001	0.0002
SSI (100)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0004)	(0.0004)
Strict Regulation	0.0001	0.0001	0.0000	0.0000	-0.0000	-0.0001
	(0.0002)	(0.0002)	(0.0002)	(0.0002)	(0.0003)	(0.0003)
Republican Governor				-		
	-0.0001**	-0.0001**	-0.0001**	0.0001***	0.0000	0.0000
	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0001)	(0.0001)
Max TANF Benefits	-0.0000	-0.0000	-0.0000	-0.0000	-0.0000	-0.0000
(Hundreds)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0001)
Medicaid Buy-In	0.0000	0.0000	-0.0000	0.0000	0.0001	0.0000
	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0001)
Maximum Weekly UI Benefits	-0.0000	-0.0000	-0.0001	-0.0001	-0.0001	-0.0001
(Hundreds)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0002)	(0.0002)
Minimum Wage	0.0000	-0.0000	0.0000	-0.0000	0.0002	0.0002
	(0.0001)	(0.0001)	(0.0000)	(0.0000)	(0.0001)	(0.0002)
2000-2003	0.0000	-0.0003*	-0.0001	-0.0001	-0.0000	-0.0004
	(0.0001)	(0.0002)	(0.0001)	(0.0002)	(0.0001)	(0.0004)
2004-2007	0.0001	0.0001	0.0000	0.0002	-0.0001	0.0002
	(0.0001)	(0.0003)	(0.0001)	(0.0002)	(0.0002)	(0.0004)
2008-2011	0.0004***	0.0001	0.0004***	0.0001	-0.0003	0.0000
	(0.0001)	(0.0002)	(0.0001)	(0.0002)	(0.0003)	(0.0003)
2000-2003*Unemployment		0.0062		0.0015		0.0103
		(0.0039)		(0.0039)		(0.0112)
2004-2007*Unemployment		0.0016		-0.0027		-0.0042
		(0.0044)		(0.0041)		(0.0056)
2008-2011*Unemployment		0.0074**		0.0056		-0.0028
		(0.0036)		(0.0035)		(0.0046)
Constant	0.0083***	0.0088***	0.0075***	0.0078***	0.0020	0.0025
	(0.0015)	(0.0014)	(0.0013)	(0.0012)	(0.0024)	(0.0022)
State Dummies	Yes	Yes	Yes	Yes	Yes	Yes
Observations	816	816	816	816	816	816
R-squared	0.9212	0.9226	0.9168	0.9185	0.8389	0.8412
Number of Clusters	51	51	51	51	51	51

Table 8. State-Level Regressions; SSI 1st Awards by Age Group (cont'd)

	Exit for Still		Exit for		Entry for	
	Eligible		Ineligible		Newly Eligible	
Mean Hazard Rate	0.031		0.016		0.065	
Age	-0.001	***	-0.002	***	-0.0001	
	(0.000)		(0.000)		(0.000)	
Female	-0.029	***	-0.001		0.026	***
	(0.002)		(0.003)		(0.001)	
Marital Status	0.036	***	0.033	***	-0.024	***
	(0.002)		(0.003)		(0.001)	
Race						
Black	-0.004	*	-0.008	***	0.018	***
	(0.002)		(0.003)		(0.002)	
Other	0.003		-0.007	*	0.003	
	(0.004)		(0.004)		(0.002)	
Education						
Less than HS	-0.001		-0.016	***	-0.0003	
	(0.002)		(0.002)		(0.001)	
College	0.010	***	0.010	***	-0.009	***
	(0.002)		(0.002)		(0.001)	
Household Size	0.003	***	0.007	***	-0.0003	
	(0.001)		(0.001)		(0.000)	
Kids	-0.011	***	-0.023	***	0.008	***
	(0.003)		(0.003)		(0.002)	
Live in an MSA	0.003		0.005	*	-0.009	***
	(0.002)		(0.003)		(0.001)	
Employment	0.026	***	0.036	***	-0.010	***
	(0.002)		(0.003)		(0.001)	
Poor Health	-0.002		-0.013	***	0.022	***
	(0.002)		(0.002)		(0.002)	
Income	0.004	***	0.021	***	-0.001	***
	(0.001)		(0.001)		(0.000)	
Own Home	0.007	***	0.026	***	-0.026	***
	(0.002)		(0.002)		(0.001)	
Own a Vehicle	0.012	***	0.036	***	-0.015	***
	(0.002)		(0.003)		(0.001)	
Receive Other Welfare	-0.059	***	-0.074	***	0.086	***
	(0.002)		(0.001)		(0.003)	
State Level Characteristics						
Unemployment rate	0.001		-0.001		0.002	***
	(0.001)		(0.001)		(0.000)	
Minimum wage	-0.002	*	-0.0002		-0.001	
	(0.001)		(0.001)		(0.001)	
Maximum TANF benefit	0.005		0.005		-0.001	
	(0.003)		(0.004)		(0.002)	
SNAP short certification	0.029	***	0.010		-0.021	***
	(0.011)		(0.013)		(0.007)	
Expanded categorical eligibility	-0.002		-0.004		0.003	*
	(0.003)		(0.004)		(0.002)	
Ν	400,618		493,228		167,128	

Table 9. Probit Regression Results for SNAP Participation.

Table 10. Probit R	egression R	esults f	or SNAP Part	icipati	ion by Age	Group	9.					
			Under 60				60 and Over					
	Exit for Still Eligible		Exit for Ineligible		Entry for Newly Eligible		Exit for Still Eligible		Exit for Ineligible		Entry for Newly Eligible	
Mean Hazard Rate	0.034		0.018		0.070		0.018		0.006		0.034	
Age	-0.001	***	-0.001	***	0.0002		-0.001		-0.001		-0.001	**
e	(0.000)		(0.000)		(0.000)		(0.000)		(0.001)		(0.000)	
Female	-0.036	***	0.000		0.031	***	-0.005		-0.002		0.005	*
	(0.002)		(0.003)		(0.002)		(0.003)		(0.004)		(0.003)	
Marital Status	0.041	***	0.038	***	-0.027	***	0.026	***	0.017	***	-0.020	***
	(0.003)		(0.003)		(0.001)		(0.004)		(0.005)		(0.002)	
Race												
Black	-0.007	**	-0.009	***	0.020	***	0.005		-0.003		0.010	***
	(0.003)		(0.003)		(0.002)		(0.004)		(0.004)		(0.004)	
Other	0.003		-0.007		0.002		-0.001		-0.017	***	0.013	**
	(0.004)		(0.005)		(0.003)		(0.005)		(0.006)		(0.006)	
Education												
Less than HS	-0.003		-0.019	***	-0.0024		0.002		-0.008	**	0.013	***
	(0.003)		(0.003)		(0.002)		(0.004)		(0.004)		(0.004)	
College	0.011	***	0.012	***	-0.012	***	0.006		-0.002		-0.001	
	(0.003)		(0.003)		(0.001)		(0.005)		(0.005)		(0.003)	
Household Size	0.003	***	0.007	***	-0.0003		0.005	***	0.012	***	-0.001	
	(0.001)		(0.001)		(0.000)		(0.001)		(0.002)		(0.001)	
Kids	-0.012	***	-0.027	***	0.008	***	-0.001		-0.014	**	-0.006	*
	(0.003)		(0.003)		(0.002)		(0.006)		(0.006)		(0.004)	
Live in an MSA	0.003		0.006	*	-0.011	***	0.006		0.002		-0.0003	
	(0.003)		(0.003)		(0.002)		(0.004)		(0.004)		(0.003)	
Employment	0.027	***	0.040	***	-0.011	***	0.021	***	0.024	***	-0.010	***
	(0.003)		(0.003)		(0.001)		(0.008)		(0.007)		(0.003)	
Poor Health	-0.005	**	-0.018	***	0.026	***	0.002		-0.002		0.013	***
	(0.003)		(0.003)		(0.002)		(0.003)		(0.004)		(0.003)	
Income	0.005	***	0.024	***	-0.002	***	0.004	***	0.024	***	-0.002	***
	(0.001)		(0.001)		(0.000)		(0.001)		(0.004)		(0.001)	
Own Home	0.007	***	0.032	***	-0.026	***	0.003		0.001		-0.025	***
	(0.003)		(0.003)		(0.001)		(0.003)		(0.004)		(0.002)	
Own a Vehicle	0.015	***	0.041	***	-0.016	***	0.001		0.018	***	-0.010	***
	(0.002)		(0.003)		(0.001)		(0.003)		(0.004)		(0.002)	

Table 10. Probit Regression Results for SNAP Participation by Age Group.

	0		Under 6	0	0 1	,	60 and Over						
	Exit for Still		Exit for		Entry for Newly		Exit for Still		Exit for		Entry for Newly		
	Eligible		Ineligible		Eligible		Eligible		Ineligible		Eligible		
Receive Other Welfare	-0.071	***	-0.086	***	0.087	***	-0.023	***	-0.037	***	0.083	***	
	(0.002)		(0.002)		(0.003)		(0.002)		(0.001)		(0.007)		
State Level Characteristics													
Unemployment rate	-0.0001		-0.002	**	0.003	***	0.002		0.003	*	-0.002	*	
	(0.001)		(0.001)		(0.001)		(0.001)		(0.002)		(0.001)		
Minimum wage	-0.002	*	-0.0003		-0.001		-0.001		-0.001		-0.001		
	(0.001)		(0.001)		(0.001)		(0.001)		(0.002)		(0.001)		
Maximum TANF benefit	0.008	**	0.006		-0.002		-0.005		-0.001		0.003		
	(0.004)		(0.004)		(0.002)		(0.005)		(0.006)		(0.004)		
SNAP short certification	0.032	**	0.007		-0.021	***	0.034	**	0.042	*	-0.027	*	
	(0.013)		(0.015)		(0.008)		(0.016)		(0.024)		(0.014)		
Categorical eligibility	0.000		-0.005		0.003		-0.009	**	0.001		0.005		
	(0.004)		(0.004)		(0.002)		(0.004)		(0.006)		(0.004)		
Ν	315,299		393,499		141,262		84,983		99,061		25,810		

Table 10. Probit Regression Results for SNAP Participation by Age Group (cont'd).

	Exit for				Entry for	
	Still		Exit for		Newly	
	Eligible		Ineligible		Eligible	
Mean Hazard Rate	0.017		0.011		0.020	
Age	-0.002	***	-0.0003		0.0003	***
	(0.000)		(0.000)		(0.000)	
Female	0.037	***	0.015	**	-0.001	
	(0.006)		(0.006)		(0.001)	
Marital Status	0.037	***	0.024	***	-0.009	***
	(0.006)		(0.007)		(0.001)	
Race						
Black	0.025	***	0.006		0.004	***
	(0.006)		(0.008)		(0.001)	
Other	0.002		-0.014		0.005	***
	(0.009)		(0.012)		(0.002)	
Education						
Less than HS	0.008		-0.006		0.002	*
	(0.006)		(0.007)		(0.001)	
College	0.010		0.008		-0.004	***
	(0.007)		(0.008)		(0.001)	
Household Size	0.007	***	0.003		-0.001	**
	(0.002)		(0.002)		(0.000)	
Kids	0.073	***	0.042	***	-0.023	***
	(0.007)		(0.008)		(0.001)	
Live in an MSA	0.009		0.007		-0.001	
	(0.007)		(0.008)		(0.001)	
Employment	0.017	*	-0.001		-0.026	***
	(0.009)		(0.009)		(0.001)	
Poor Health	-0.018	***	0.016	**	0.021	***
	(0.005)		(0.006)		(0.001)	
Own Home	-0.009		-0.009		-0.001	
	(0.005)		(0.007)		(0.001)	
Own a Vehicle	-0.003		0.012	*	-0.007	***
	(0.005)		(0.007)		(0.001)	
Receive Other Welfare	-0.286	***	-0.304	***	0.114	***
	(0.003)		(0.002)		(0.006)	
State Level Characteristics						
Unemployment rate	0.007	***	0.004		0.0000	
	(0.002)		(0.003)		(0.000)	
Minimum wage	-0.004	*	-0.005	**	-0.0001	
	(0.003)		(0.003)		(0.001)	
Maximum TANF benefit	0.009		0.015		0.002	
	(0.009)		(0.010)		(0.002)	
Maximum SSI benefit	-0.029		0.005		0.005	
	(0.025)		(0.031)		(0.005)	
Ν	253,123		305,470		192,182	

Table 11. Probit Regression Results for SSI Participation.

	0	0	Under 65	, v			65 and Over					
	Exit for				Entry for		Exit for				Entry for	
	Still		Exit for		Newly		Still		Exit for		Newly	
	Eligible		Ineligible		Eligible		Eligible		Ineligible		Eligible	
Mean Hazard Rate	0.020		0.012		0.016		0.012		0.009		0 104	
Age	-0.002	***	0 0005		0 0004	***	-0.0003		-0.002		0 0000	
	(0.000)		(0.001)		(0.000)		(0.002)		(0.002)		(0.001)	
Female	0.046	***	0.011		-0.001		-0.006		0.044	***	0.002	
	(0.006)		(0.007)		(0.001)		(0.012)		(0.016)		(0.004)	
Marital Status	0.035	***	0.026	***	-0.007	***	0.026	*	0.039	**	-0.016	***
	(0.007)		(0.008)		(0.001)		(0.013)		(0.017)		(0.004)	
Race							· · · ·		· · · ·		~ /	
Black	0.031	***	0.006		0.003	**	-0.003		0.014		0.006	
	(0.007)		(0.008)		(0.001)		(0.013)		(0.016)		(0.005)	
Other	0.005		-0.017		0.001		0.002		0.008		0.019	***
	(0.012)		(0.014)		(0.002)		(0.017)		(0.025)		(0.007)	
Education												
Less than HS	0.006		-0.0005		0.001		0.020		-0.010		0.016	***
	(0.007)		(0.008)		(0.001)		(0.014)		(0.016)		(0.005)	
College	0.017	**	0.014	*	-0.004	***	-0.012		-0.016		0.005	
	(0.008)		(0.008)		(0.001)		(0.018)		(0.020)		(0.005)	
Household Size	0.006	***	0.003		-0.001	***	0.012	***	-0.007		0.004	**
	(0.002)		(0.002)		(0.000)		(0.004)		(0.006)		(0.001)	
Kids	0.074	***	0.036	***	-0.020	***	0.039	*	0.080	***	-0.020	***
	(0.008)		(0.009)		(0.001)		(0.020)		(0.025)		(0.005)	
Live in an MSA	-0.0004		0.002		-0.001		0.043	***	0.015		-0.004	
	(0.008)		(0.009)		(0.001)		(0.015)		(0.020)		(0.005)	
Employment	0.024	***	0.010		-0.024	***	-0.076	**	-0.060	**	-0.017	***
	(0.009)		(0.009)		(0.001)		(0.036)		(0.029)		(0.006)	
Poor Health	-0.021	***	0.019	***	0.021	***	-0.009		0.001		0.007	*
	(0.006)		(0.007)		(0.001)		(0.011)		(0.014)		(0.004)	
Income	0.028	***	0.148	***	-0.0003		0.046	***	0.298	***	-0.007	***
	(0.002)		(0.010)		(0.000)		(0.004)		(0.025)		(0.001)	
Own Home	-0.009		-0.009		0.0004		0.003		-0.013		-0.022	***
0 1111	(0.006)		(0.007)		(0.001)		(0.012)		(0.015)	-11-	(0.004)	
Own a Vehicle	-0.005		0.005		-0.006	***	0.010		0.040	**	-0.008	**
	(0.006)		(0.008)		(0.001)		(0.012)		(0.016)		(0.004)	

Table 12. Probit Regression Results for SSI Participation by Age Group.

	v		Under 6	5			65 and Over						
	Exit for				Entry for		Exit for				Entry for		
	Still		Exit for		Newly		Still		Exit for		Newly		
	Eligible		Ineligible		Eligible		Eligible		Ineligible		Eligible		
Receive Other Welfare	-0.288	***	-0.281	***	0.079	***	-0.292	***	-0.395	***	0.234	***	
	(0.004)		(0.003)		(0.005)		(0.004)		(0.004)		(0.025)		
State Level Characteristics													
Unemployment rate	0.005	*	0.003		0.0001		0.015	***	0.007		-0.001		
	(0.003)		(0.003)		(0.000)		(0.005)		(0.006)		(0.002)		
Minimum wage	-0.002		-0.007	**	-0.0002		-0.014	***	0.002		0.003		
	(0.003)		(0.003)		(0.000)		(0.005)		(0.006)		(0.002)		
Maximum TANF benefit	0.003		0.015		0.002		0.030		0.010		-0.008		
	(0.010)		(0.011)		(0.002)		(0.020)		(0.026)		(0.006)		
Maximum SSI benefit	-0.033		-0.011		0.008	*	-0.036		0.066		-0.012		
	(0.030)		(0.035)		(0.005)		(0.048)		(0.067)		(0.017)		
N	177,044		213,645		183,613		75,642		91,665		8,424		

Table 12. Probit Regression Results for SSI Participation by Age Group (cont'd).

	Exit for				Entry for	
	Still		Exit for		Newly	
	Eligible		Ineligible		Eligible	
Employment	0.028	***	0.039	***	-0.010	***
	(0.005)		(0.005)		(0.002)	
Employment and Period 2	-0.026	***	-0.030	***	0.002	
	(0.009)		(0.010)		(0.003)	
Employment and Period 3	-0.040	***	-0.033	***	-0.003	
	(0.008)		(0.009)		(0.003)	
Employment and Period 4	-0.053	***	-0.052	***	-0.008	***
	(0.007)		(0.008)		(0.003)	
Poor Health	-0.002		-0.012	***	0.023	***
	(0.004)		(0.004)		(0.003)	
Poor Health and Period 2	-0.017	**	0.004		0.004	
	(0.007)		(0.010)		(0.003)	
Poor Health and Period 3	-0.012	*	-0.010		0.002	
	(0.006)		(0.009)		(0.003)	
Poor Health and Period 4	-0.020	***	-0.015	*	0.004	
	(0.006)		(0.008)		(0.004)	
Income	0.004	***	0.021	***	-0.001	**
	(0.001)		(0.002)		(0.001)	
Income and Period 2	0.003	*	0.005	**	0.0001	
	(0.002)		(0.002)		(0.001)	
Income and Period 3	0.003		0.004	**	-0.0003	
	(0.002)		(0.002)		(0.001)	
Income and Period 4	0.004	***	0.002		0.0001	
	(0.001)		(0.002)		(0.001)	
Ν	400,618		493,228		167,128	

 Table 13A. Probit Regression Results for SNAP Participation with Interactions.

	Exit for				Entry for	
	Still		Exit for		Newly	
	Eligible		Ineligible		Eligible	
Employment	0.016		-0.001		-0.026	***
	(0.017)		(0.016)		(0.001)	
Employment and Period 2	-0.002		-0.055	**	-0.002	
	(0.028)		(0.025)		(0.002)	
Employment and Period 3	0.007		-0.031		-0.002	
	(0.024)		(0.023)		(0.002)	
Employment and Period 4	0.017		-0.025		-0.004	
	(0.023)		(0.021)		(0.003)	
Poor Health	-0.018	*	0.014		0.021	***
	(0.009)		(0.012)		(0.003)	
Poor Health and Period 2	0.015		-0.021		0.003	
	(0.014)		(0.018)		(0.002)	
Poor Health and Period 3	0.004		0.027		0.004	*
	(0.014)		(0.021)		(0.002)	
Poor Health and Period 4	0.008		0.022		0.0003	
	(0.012)		(0.019)		(0.002)	
Income	0.031	***	0.167	***	-0.001	*
	(0.003)		(0.019)		(0.001)	
Income and Period 2	0.008	**	0.012		-0.0001	
	(0.004)		(0.022)		(0.001)	
Income and Period 3	0.009	**	0.005		0.002	***
	(0.004)		(0.017)		(0.001)	
Income and Period 4	0.007	*	0.001		0.001	
	(0.004)		(0.016)		(0.001)	
N	253,123		305,470		192,182	

 Table 13B. Probit Regression Results for SSI Participation with Interactions.